# **ECONSTOR** Make Your Publications Visible.

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Yousaf, Imran; Ali, Shoaib; Wong, Wing Keung

# Article

# An empirical analysis of the volatility spillover effect between world-leading and the Asian stock markets: Implications for portfolio management

Journal of Risk and Financial Management

**Provided in Cooperation with:** MDPI – Multidisciplinary Digital Publishing Institute, Basel

*Suggested Citation:* Yousaf, Imran; Ali, Shoaib; Wong, Wing Keung (2020) : An empirical analysis of the volatility spillover effect between world-leading and the Asian stock markets: Implications for portfolio management, Journal of Risk and Financial Management, ISSN 1911-8074, MDPI, Basel, Vol. 13, Iss. 10, pp. 1-28, https://doi.org/10.3390/jrfm13100226

This Version is available at: https://hdl.handle.net/10419/239330

### Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.



WWW.ECONSTOR.EU

https://creativecommons.org/licenses/by/4.0/

#### Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.





Article



# An Empirical Analysis of the Volatility Spillover Effect between World-Leading and the Asian Stock Markets: Implications for Portfolio Management

Imran Yousaf <sup>1,\*</sup>, Shoaib Ali <sup>1</sup> and Wing-Keung Wong <sup>2,3,4</sup>

- <sup>1</sup> Air University School of Management, Air University, Islamabad 44000, Pakistan; ShoaibAli@mail.au.edu.pk
- <sup>2</sup> Department of Finance, Fintech Center, and Big Data Research Center, Asia University, Taichung 41354, Taiwan; wong@asia.edu.tw
- <sup>3</sup> Department of Medical Research, China Medical University Hospital, Taichung 40402, Taiwan
- <sup>4</sup> Department of Economics and Finance, The Hang Seng University of Hong Kong, Hong Kong 999077, China
- \* Correspondence: imranyousaf.fin@gmail.com

Received: 14 August 2020; Accepted: 23 September 2020; Published: 25 September 2020



Abstract: This study employs the Vector Autoregressive-Generalized Autoregressive Conditional Heteroskedasticity (VAR-AGARCH) model to examine both return and volatility spillovers from the USA (developed) and China (Emerging) towards eight emerging Asian stock markets during the full sample period, the US financial crisis, and the Chinese Stock market crash. We also calculate the optimal weights and hedge ratios for the stock portfolios. Our results reveal that both return and volatility transmissions vary across the pairs of stock markets and the financial crises. More specifically, return spillover was observed from the US and China to the Asian stock markets during the US financial crisis and the Chinese stock market crash, and the volatility was transmitted from the USA to the majority of the Asian stock markets during the Chinese stock market crash. Additionally, volatility was transmitted from China to the majority of the Asian stock markets during the US financial crisis. The weights of American stocks in the Asia-US portfolios were found to be higher during the Chinese stock market crash than in the US financial crisis. For the majority of the Asia-China portfolios, the optimal weights of the Chinese stocks were almost equal during the Chinese stock market crash and the US financial crisis. Regarding hedge ratios, fewer US stocks were required to minimize the risk for Asian stock investors during the US financial crisis. In contrast, fewer Chinese stocks were needed to minimize the risk for Asian stock investors during the Chinese stock market crash. This study provides useful information to institutional investors, portfolio managers, and policymakers regarding optimal asset allocation and risk management.

**Keywords:** return spillover; volatility spillover; shock spillover; US financial crisis; Chinese stock market crash

## JEL Classification: G10; G11; G12; G15

### 1. Introduction

Information transmissions from both return and volatility across national equity markets are of greater interest to both investors and policymakers, with increasing financial integration in the stock markets all over the world (Yousaf et al. 2020). If, for example, asset volatility is transmitted from one market to another during turmoil or crisis period (Forbes and Rigobon 2002; Diebold and Yilmaz 2009), then portfolio managers need to adjust their asset allocations (Baele 2005; Engle et al. 2012) and financial policymakers need to adapt their policies in order to mitigate the contagion risk. Changes in linkages

between national equity markets, especially during a crisis, can also have important implications for asset allocations, business valuation, risk management, and access to finance.

Several studies have examined linkages between the equity markets during the 1997 Asian financial crisis (In Francis et al. 2001; Wan and Wong 2001; Yang et al. 2003), and the last 2008 global financial crisis (Yilmaz 2010; Cheung et al. 2007; Kim et al. 2015; Li and Giles 2015; Lean et al. 2015; Vieito et al. 2015; Zhu et al. 2019) and some studies, see, for example, Fung et al. (2011) and Guo et al. (2017), develop theories to explain that crisis. However, the linkages between equity markets during the Chinese stock market crash of 2015have been rarely examined. The Chinese stock market experienced a major crash in 2015 (Zhu et al. 2017; Yousaf and Hassan 2019; Yousaf et al. 2020; Yousaf and Ali 2020). The CSI 300 index increased before reaching 5178 points in mid-June of 2015. Then, it took a roller-coaster ride and dropped by up to 34% in just 20 days; Chinese stock market also lost 1000 points within just one week. Around 50% of Chinese stocks lost more than half of their pre-crash market value. The Chinese stock market crash affected many other commodities and financial markets, including Asian (Allen 2015) and the US stock markets (The causes and consequences of China's market crash 2015).

Despite the importance of the Chinese crash for international portfolio managers, few studies have examined how it was transmitted to other national financial markets. Xiong et al. (2018) investigate the time-varying correlation between economic policy uncertainty and Chinese stock market returns during the Chinese crash of 2015, while Yousaf and Hassan (2019) examine the linkages between crude oil and emerging Asian stock markets during this crisis. However, research on the linkages between stock markets has not been investigated yet for the 2015 Chinese crash. Therefore, this study focuses on providing useful insights about this issue for the Asian region, which has attracted considerable attention from finance practitioners and academics due to its position as the center of global economic activity in the 21st century<sup>1</sup>. While using the US and Chinese equity markets as the indicators of global markets, we explore whether global investors can get the maximum benefit of diversification by adding emerging Asian market stocks in their portfolios. In literature, several studies have examined the linkages between the global (US and China) and emerging Asian equity markets during the Asian financial crisis, and the US financial crisis (Yang et al. 2003; Beirne et al. 2013; Jin 2015; Li and Giles 2015), but not in the Chinese stock market crash.

We address the above-mentioned literature gap by examining the return and volatility spillover from the US and China to the emerging Asian equity markets during the Chinese stock market crash by using the VAR-AGARCH model that was developed by Ling and McAleer (2003). Moreover, we examine the ability of spillovers during the full sample period and the 2008 US financial crisis to provide comparative insights to investors about whether the impact of the Chinese crash on equity market spillovers was different from those in the other sample periods. Our findings show that return spillover was observed from the US and China to the Asian stock market during the US financial crisis and the Chinese stock market crash. Volatility was also transmitted from the US to the majority of the Asian stock markets during the Chinese stock market crash. However, volatility was transmitted from China to the majority of the Asian stock markets during the US financial crisis. Overall, as the return and volatility transmission vary across pairs of stock markets and financial crises, investors have to adjust their asset allocations from time to time to improve their profits. Therefore, we also estimate the optimal weights and hedge ratios during the full sample period, the US financial crisis, and the Chinese stock market crash. Our findings imply that fewer US stocks were required to minimize the risk for Asian stock investors during the US financial crisis compared to during the Chinese crash. In contrast, fewer Chinese stocks were needed to minimize the risk for Asian stock investors during the Chinese stock market crash as compared to during the US crisis. Overall, our findings draw several important implications for risk management and portfolio diversification that could be useful for investors and policymakers related to the US and Asian stock markets.

<sup>&</sup>lt;sup>1</sup> Source: https://www.ft.com/content/520cb6f6-2958-11e9-a5ab-ff8ef2b976c7.

The rest of the paper is organized as follows: Section 2 provides the literature review. Section 3 describes the data and methodology. Section 4 reports the findings, and Section 5 concludes the whole discussion.

#### 2. Literature Review

The analysis of both return and volatility spillover between stock markets is crucial for investors in designing optimal portfolios. According to modern portfolio theory, the gains of international portfolio diversification decrease when the correlation of security returns increases and vice versa. Michaud et al. (1996) discuss the advantages of a low correlation between the developed and emerging markets for international portfolio diversification. Due to this trend, investors can benefit by investing in emerging markets that are weakly interconnected with developed markets. However, this correlation becomes higher during an economic crisis, suggesting low diversification benefits when diversification is most required.

#### 2.1. Linkages between US, China, and Asian Stock Markets

Many studies have been conducted to investigate the link between different stock markets during the last three decades. Liu and Pan (1997) examine the mean and the volatility spillover from the US and Japan to Singapore, Hong Kong, Thailand, and Taiwan. The results show that the US market is more dominant than the Japanese stock market in transmitting return and volatility effects to four Asian stock markets. Huang et al. (2000) investigate the link between the US, Japan, and South China growth triangle. The US stock market significantly and dominantly affects the south Chinese growth triangle compared to the impact of Japan on China's stock market. The return spillover has been also found to be significant from the US to Hong Kong and Taiwan, and from Hong Kong to the Taiwanese stock market. Miyakoshi (2003) estimates the return and volatility spillover between the US, Japan, and seven Asian stock markets (South Korea, Taiwan, Singapore, Thailand, Indonesia, and Hong Kong). It finds a significant return spillover from the US to Asian markets, whereas no return spillover is found from Japan to Asian stock markets. Moreover, the volatility spillover from the US to Asian stock markets is observed to be dominant as compared to the volatility spillover from the US to Asian stock markets.

Johansson and Ljungwall (2009) examine the association between stock markets of China, Hong Kong, and Thailand. It reports a significant return spillover from Taiwan to China and the Hong Kong stock market. In contrast, volatility spillover runs from Hong Kong to Taiwan and from Taiwan to the Chinese stock market. Zhou et al. (2012) estimate the spillover between Chinese and international (the US, the UK, France, Germany, Japan, India, Hong Kong, Taiwan, South Korea, and Singapore) stock markets from 1996 to 2009. Before 2005, the Chinese stock market was affected by spillover from other international markets. After 2005, volatility spillover was significantly transmitted from China to most of the other international stock markets. Chien et al. (2015) report on the significant financial integration between China and the ASEAN-5 (Indonesia, Malaysia, the Philippines, Singapore, and Thailand) stock markets. Huo and Ahmed (2017) provide significant evidence of both return and volatility effects from China to the Hong Kong equity market.

#### 2.2. Linkages between US, China, and Asian Stock Markets during Crisis

Many studies have examined the linkages between markets during crisis periods. Yang et al. (2003) investigate the short and long-run relationship between the US, Japan, and ten Asian stock markets, mainly focusing on the Asian financial crisis of 1997–1998. This study reports a strengthened long-run co-integration among these stock markets during the Asian financial crises. The degree of integration is found to change during crises and non-crisis periods. Beirne et al. (2013) look at the volatility spillover from developed to emerging stock markets during periods of turbulence in mature stock markets. It finds that volatility in mature markets affects the conditional variances in emerging stock markets.

Moreover, the spillover effect from developed to emerging markets is also changed during times of turbulence in mature markets.

Jin (2015) examines the mean and volatility spillover between China, Taiwan, and Hong Kong. It reveals that financial crises have a substantial and positive effect on expected conditional variances, but also that the size and dynamics of impacts vary from market to market. Li and Giles (2015) investigate the volatility spillover across the US, Japan, and four Asian developing economies during the Asian financial crisis of 1997 and the US financial crises of 2008. The results revealed that there is a presence of a volatility spillover effect from the USA to Asian developing economies and Japan. This study also finds a bidirectional volatility spillover between US and Asian markets that occurred during the Asian financial crisis. Gkillas et al. (2019) explore integration and co-movement between 68 international stock markets (including in the Asian region) during the US financial crisis.

Overall, several studies have examined the return and volatility spillover from the US to Asian markets during the Asian financial crisis of 1997 and the US financial crisis of 2008. However less has been done on both return and volatility transmission from China to the emerging Asian stock markets during the US financial crisis and the Chinese stock market crash. Moreover, no study has examined return and volatility spillovers from the US to the emerging Asian stock markets during the Chinese crash. Therefore, this study addresses these above-mentioned literature gaps.

#### 3. Data and Methodology

#### 3.1. Data

We based our empirical investigation on daily data of accepted benchmark stock indices of nine Asian countries and the US. The Emerging Asian stock markets include China, India, South Korea, Indonesia, Pakistan, Malaysia, the Philippines, Thailand, and Taiwan. The emerging Asian economies were selected from the list of countries, including the MSCI (Morgan Stanley Capital International) emerging market index. The data of stock indices were taken from the Data Stream database. The index is assumed to be the same on non-trading days (holidays except weekends) as on the previous trading day, as suggested by Malik and Hammoudeh (2007) and many others.<sup>2</sup>

This study used the full sample period from 1 January 2000 to 30 June 2018 and studies the following two sub-samples: the first sub-period from 1 August 2007 to 31 July 2010 presenting the period with the US financial crisis; and the second sub-period from 1 June 2015 to 30 May 2018 presenting the period with the Chinese Stock market crash. We note that Yousaf and Hassan (2019) also use similar time frames for the US financial crisis and the Chinese stock market crash. This study followed He (2001) and many others to use three-year data for each crisis for short-run analysis. Changes in market correlations take place continuously not only as a result of crises but also due to the consequences of many financial, economic, and political events. Moreover, Arouri et al. (2015) have also used the daily data covering periods shorter than three years to estimate the return and volatility spillover between gold and Chinese stock markets in US financial crisis by applying the VAR-GARCH model. The difference in the opening time of US and Asian stock markets was adjusted by using lags where necessary.

<sup>&</sup>lt;sup>2</sup> In time-series data, if there are missing values, there are two ways to deal with the incomplete data: (a) omit the entire record that contains information, (b) Impute the missing information. We used 10 series in this paper and if we wanted to omit the missing data for one series then the data of all other nine series needed to be removed as well for that specific day. So, if we omitted the data for days where values are missing at specific days, then we lost the data for many days, which is not good for getting realistic results. Therefore, we followed many studies, for example, Malik and Hammoudeh (2007), and imputed the missing data by using previous day data. Indeed, there are many methods used to impute the missing data and every method has pros and cons, but we used this imputation method following past literature. Moreover, our missing observations were less than one percent of overall data, therefore the imputation method should not create a larger effect than that on results.

#### 3.2. Methodology

This study estimated the return and volatility transmissions using the Vector Autoregressive-Generalized Autoregressive Conditional Heteroskedasticity (VAR-AGARCH) model proposed by McAleer et al. (2009). Several studies have previously used the VAR-GARCH and VAR-AGARCH model to estimate spillover between different asset classes (Arouri et al. 2011; Arouri et al. 2012; Jouini 2013; Yousaf and Hassan 2019). This model includes the Constant Conditional Correlation (CCC-GARCH) model of Bollerslev (1990) as a special case. The selection of the model was based on three reasons. First, the most commonly used multivariate models are the BEKK (Baba, Engle, Kraft, and Kroner) model and the DCC (dynamic conditional correlation) model. These models often suffer from unreasonable parameter estimates and data convergence problems (Bouri 2015). The VAR-AGARCH model overcomes these problems regarding parameters and data convergence. Second, it incorporates asymmetry into the model. Third, this model can be used to calculate the optimal weights and hedge ratios.

Ling and McAleer (2003) propose the multivariate VAR-GARCH Model to estimate the return and volatility transmission between the different series. For two series, the VAR-GARCH model has the following specifications for the conditional mean equation<sup>3</sup>:

$$R_t = \mu + FR_{t-1} + e_t \text{ with } e_t = D_t^{1/2} \eta_t,$$
(1)

in which  $R_t$  represents a 2 × 1 vector of daily returns<sup>4</sup> on the stocks *x* and *y* at time *t*,  $\mu$  denotes a 2 × 1 vector of constants, *F* is a 2 × 2 matrix of parameters measuring the impacts of own lagged and cross mean transmissions between two series,  $e_t$  is the residual of the mean equation for the two stocks returns series at time *t*,  $\eta_t$  indicates a 2 × 1 vector of independently and identically distributed random vectors, and  $D_t^{1/2} = diag (\sqrt{h_t^x}, \sqrt{h_t^y})$ , where  $h_t^x$  and  $h_t^y$  representing the conditional variances of the returns for stocks x and y, respectively, are given as

$$h_t^x = C_x + a_{11} \left( e_{t-1}^x \right)^2 + a_{21} \left( e_{t-1}^y \right)^2 + b_{11} h_{t-1}^x + b_{21} h_{t-1'}^y$$
(2)

$$h_t^y = C_y + a_{12} \left( e_{t-1}^x \right)^2 + a_{22} \left( e_{t-1}^y \right)^2 + b_{12} h_{t-1}^x + b_{22} h_{t-1}^y.$$
(3)

Equations (2) and (3) reveal how shock and volatility are transmitted over time and across the markets under investigation. Furthermore, the conditional covariance between returns from two different stock markets can be estimated as follows:

$$h_t^{x,y} = p \times \sqrt{h_t^x} \times \sqrt{h_t^y}.$$
(4)

In the above equation,  $h_t^{x,y}$  refers to the conditional covariance between the returns of two stock markets (*x*, *y*) at time *t*. Moreover, *p* indicates the constant conditional correlation between the returns of two stock markets (*x*, *y*).

The VAR–GARCH model assumes that positive or negative shocks have the same impact on the conditional variance. To estimate the spillover between different markets, we estimated spillover between two stock markets by using the VAR–AGARCH Model proposed by the McAleer et al. (2009).

<sup>&</sup>lt;sup>3</sup> Several studies, for example, Hammoudeh et al. (2009), Arouri et al. (2011), and Dutta et al. (2018) have applied the VAR for the conditional mean equation.

<sup>&</sup>lt;sup>4</sup> Stock Returns<sub>t</sub> =  $ln\left(\frac{Stock Index_t}{Stock Index_{t-1}}\right)$ .

The VAR AGARCH model incorporates asymmetry in the model as well. Specifically, instead of using Equations (2) and (3), the conditional variance of the VAR AGARCH model was defined as follows:

$$h_t^x = C_x + a_{11}A(e_{t-1}^x)^2 + a_{21}A(e_{t-1}^y)^2 + b_{11}h_{t-1}^x + b_{21}h_{t-1}^y + a_{11}B[(e_{t-1}^x)((e_{t-1}^x) < 0))],$$
(5)

$$h_t^y = C_y + a_{12}A(e_{t-1}^x)^2 + a_{22}A(e_{t-1}^y)^2 + b_{12}h_{t-1}^x + b_{22}h_{t-1}^y + a_{22}B[(e_{t-1}^y)((e_{t-1}^y) < 0))].$$
(6)

In the above equations,  $A(e_{t-1}^x)^2$  and  $B[(e_{t-1}^x)((e_{t-1}^x) < 0))]$  as well as  $A(e_{t-1}^y)^2$  and  $B[(e_{t-1}^y)((e_{t-1}^y) < 0))]$  reveal the relationships between a market's volatility and both positive and negative own lagged returns, respectively (Lin et al. 2014). Equations (5) and (6) show that the conditional variance of each market depends upon its past shock and past volatility, as well as the past shock and past volatility of other markets. In Equation (5),  $(e_{t-1}^x)^2$  and  $(e_{t-1}^y)^2$  explain how the past shocks of both x and y affect the current conditional volatility of x. Moreover,  $h_{t-1}^x$  and  $h_{t-1}^y$  measure how the past volatilities of both x and y affect the current conditional volatility of x. The parameters of the VAR-AGARCH model can be estimated by using the Quasi-Maximum Likelihood estimation (QMLE) and using the BFGS algorithm.<sup>5</sup>

The estimates of the VAR-AGARCH model can be used to calculate optimal portfolio weights. This study followed Kroner and Ng (1998) to calculate the optimal portfolio weights for the pairs of the stock market (x, y) as:

$$w_{xy,t} = \frac{h_{y,t} - h_{xy,t}}{h_{x,y} - 2h_{xy,t} + h_{y,t}}$$

$$w_{xy,t} = \begin{cases} 0 & if \ W_{xy,t} < 0 \\ w_{xy,t} & if \ 0 \le w_{xy,t} \le 1, \\ 1 & if \ w_{xy,t} > 1 \end{cases}$$
(7)

where  $w_{xy,t}$  is the weight of stock(x) in a \$1 stock(x)-stock(y) portfolio at time t,  $h_{xy,t}$  is the conditional covariance between the two stock markets,  $h_{x,t}$  and  $h_{y,t}$  are the conditional variance of stock(x) and stock(y), respectively, and 1- $w_{xy,t}$  is the weight of stock(y) in a \$1 stock(x)-stock(y) portfolio.

It is also essential to estimate the risk-minimizing optimal hedge ratios for the portfolio of different stocks. The estimates of the VAR-AGARCH model can also be used to calculate optimal hedge ratios. This study followed Kroner and Sultan (1993) to calculate the optimal hedge ratios as:

$$\beta_{xy,t} = \frac{h_{xy,t}}{h_{y,t}},\tag{8}$$

where  $\beta_{xy,t}$  represents the hedge ratio. This shows that a short position in the stock (*y*) market can hedge a long position in the stock (*x*). Lastly, RATS 10.0 software is used for estimations.

#### 4. Empirical Results

#### 4.1. Descriptive Analysis

Table 1 reports the summary statistics of the daily returns for the US, China, and eight emerging Asian stock markets, namely India, Korea, Indonesia, Pakistan, Malaysia, the Philippines, Thailand, and Taiwan. The average returns of the Pakistani stock market are the highest out of these markets, whereas the lowest returns are found in the US stock market during the full sample period. The unconditional volatility is lower in Malaysia and the US market and is highest in the

<sup>&</sup>lt;sup>5</sup> Arouri et al. (2011), Sadorsky (2012), and Allen et al. (2013) use the Quasi-Maximum Likelihood estimation (QMLE) and use the BFGS algorithm to estimate the parameters in the VAR-GARCH model.

Chinese stock market. The skewness is negative in all cases, kurtosis is higher than 3 for all stocks, and Jarque–Bera statistics do not accept the hypothesis of the normality for all stocks. Moreover, we applied the Ljung–Box Q test for autocorrelation to the standardized residuals and squared standardized residuals. The coefficients both Q(12) and  $Q^2(12)$  were found to be significant for all series. ARCH effects were also statistically significant for all series.<sup>6</sup>

	Mean	Median	Max	Min	Std. Dev.	Skewness	Kurtosis	Jarque-Bera	Q-Stat	ARCH Test
USA	0.00016	0.00055	0.10958	-0.09470	0.01200	-0.20353	11.57202	14802.7 <sup>a</sup>	37.24 <sup>a</sup>	206.42 <sup>a</sup>
CHN	0.00045	0.00096	0.09401	-0.09256	0.01570	-0.31725	8.21506	5547.4 <sup>a</sup>	54.64 <sup>a</sup>	180.10 <sup>a</sup>
IND	0.00050	0.00094	0.15990	-0.11809	0.01472	-0.22234	10.54239	11474.1 <sup>a</sup>	84.62 <sup>a</sup>	283.89 <sup>a</sup>
INDO	0.00046	0.00113	0.07623	-0.10954	0.01357	-0.85402	10.92376	13206.3 <sup>a</sup>	154.0 <sup>a</sup>	457.66 <sup>a</sup>
KOR	0.00028	0.00080	0.11284	-0.12805	0.01509	-0.57337	9.64860	9149.3 <sup>a</sup>	24.06 a	210.02 <sup>a</sup>
MYS	0.00020	0.00041	0.04503	-0.09979	0.00816	-0.85496	13.33067	22038.9 <sup>a</sup>	226.8 <sup>a</sup>	267.36 <sup>a</sup>
PAK	0.00081	0.00109	0.08507	-0.07741	0.01359	-0.34875	6.83764	3058.01 <sup>a</sup>	165.5 <sup>a</sup>	594.62 <sup>a</sup>
PHL	0.00038	0.00055	0.16178	-0.13089	0.01309	0.23024	19.78304	56658.3 <sup>a</sup>	96.40 <sup>a</sup>	161.15 <sup>a</sup>
TAIW	0.00018	0.00070	0.06525	-0.09936	0.01356	-0.27454	6.59593	2659.6 <sup>a</sup>	77.68 <sup>a</sup>	201.54 <sup>a</sup>
THA	0.00044	0.00064	0.10577	-0.16063	0.01316	-0.70520	12.86191	19948.5 <sup>a</sup>	70.19 <sup>a</sup>	656.27 <sup>a</sup>

Table 1. Summary Statistics.

Notes: <sup>a</sup> indicates the statistical significance at 1% level.

#### 4.2. Return, Shock and Volatility Spillover Analysis

#### 4.2.1. Stock Market Linkages between the USA and Asia from the Full Sample Period

Table 2 represents the return and volatility spillover between US and Asian stock markets during the full sample period. The lagged stock returns were found to significantly affect the current stock returns in all studied Asian stock markets except for Korea. This highlights the possibility of short-term predictions of current returns through past returns in the Asian stock markets. Moreover, the autoregressive term of the USA stock market was found to be significant as well. This depicts that past returns help to predict current returns in the American stock market.

The estimate of return spillover from one market to another market can be estimated by using the coefficient of lagged return of one market (i.e., the US) onto another market (i.e., India) and vice versa. The return spillover from the USA to all Asian stock markets is significant. This implies that US stock market prices play an important role in predicting the prices of all Asian stock markets during the full sample period. These results are in line with the findings of Huyghebaert and Wang (2010), which find a significant return spillover from the USA to Asian markets. This shows that the effect of the returns of the American stock market are significantly transmitted to the Asian stock markets. However, the return spillover from all Asian stock markets to the USA was found to be insignificant. This implies that Asian stock market prices are not helpful in predicting the prices of the US stock market during the full sample period.

<sup>&</sup>lt;sup>6</sup> We applied both Augmented Dickey–Fuller (ADF), and Phillip–Perron (PP) tests to examine the stationarity of all returns series and found that all returns series are stationary, but we do not report these results in Table form for the sake of brevity.

	IND	USA	INDO	USA	KOR	USA	MYS	USA	PAK	USA	PHL	USA	TAIW	USA	THA	USA
							Panel A	: Mean Equa	ition							
Constant	$4.90 \times 10^{-4 a}$ (0.001)	$2.48 \times 10^{-4 \text{ c}}$ (0.023)	$5.13 \times 10^{-4 a}$ (0.001)	$2.40 \times 10^{-4 b}$ (0.022)	$1.55 \times 10^{-4}$ (0.252)	$2.08 \times 10^{-4 c}$ (0.059)	$1.08 \times 10^{-4}$ (0.195)	$2.46 \times 10^{-4 b}$ (0.026)	$8.79 \times 10^{-4 a}$ (0.000)	$2.07 \times 10^{-4 b}$ (0.048)	$3.53 \times 10^{-4 \text{ b}}$ (0.030)	$2.20 \times 10^{-4 a}$ (0.001)	$1.28 \times 10^{-4}$ (0.296)	$2.64 \times 10^{-4 b}$ (0.016)	$6.11 \times 10^{-4 a}$ (0.000)	$2.71 \times 10^{-4 c}$ (0.010)
$r_{t-1}^o$	0.100 <sup>a</sup> (0.000)	$-7.65 \times 10^{-4}$ (0.933)	0.136 <sup>a</sup> (0.000)	-0.010 (0.260)	0.017 (0.241)	0.016 (0.126)	0.168 <sup>a</sup> (0.000)	-0.022 (0.232)	0.168 <sup>a</sup> (0.000)	$-4.52 \times 10^{-3}$ (0.648)	0.118 <sup>a</sup> (0.000)	$-1.87 \times 10^{-3}$ (0.132)	0.054 <sup>a</sup> (0.000)	0.014 (0.232)	0.093 <sup>a</sup> (0.000)	$-8.99 \times 10^{-3}$ (0.423)
$r_{t-1}^u$	0.231 <sup>a</sup> (0.000)	-0.041 <sup>a</sup> (0.009)	0.301 <sup>a</sup> (0.000)	-0.032 <sup>b</sup> (0.031)	0.420 <sup>a</sup> (0.000)	-0.037 <sup>b</sup> (0.017)	0.201 <sup>a</sup> (0.000)	-0.034 <sup>b</sup> (0.035)	$6.40 \times 10^{-3 b}$ (0.017)	-0.030 <sup>b</sup> (0.040)	0.421 <sup>a</sup> (0.000)	-0.030 <sup>b</sup> (0.014)	0.386 <sup>a</sup> (0.000)	-0.037 <sup>b</sup> (0.021)	0.224 <sup>a</sup> (0.000)	-0.037 <sup>b</sup> (0.021)
							Panel B:	Variance Equ	ation							
Constant	$2.68 \times 10^{-6 a}$ (0.000)	$1.72 \times 10^{-6 a}$ (0.000)	$1.46 \times 10^{-5 a}$ (0.000)	$1.47 \times 10^{-6 a}$ (0.000)	$1.40 \times 10^{-6 a}$ (0.000)	$1.71 \times 10^{-6 a}$ (0.000)	$6.67 \times 10^{-7 a}$ (0.000)	$1.70 \times 10^{-6 a}$ (0.000)	$7.61 \times 10^{-6 a}$ (0.000)	$2.30 \times 10^{-6 a}$ (0.000)	$3.82 \times 10^{-5 a}$ (0.000)	$-1.46 \times 10^{-6 a}$ (0.000)	$1.11 \times 10^{-6 a}$ (0.000)	1.64 × 10 <sup>-6 a</sup> (0.000)	$4.90 \times 10^{-6 a}$ (0.000)	$1.72 \times 10^{-6 a}$ (0.000)
$\left(e^{o}_{t-1}\right)^{2}$	0.056 <sup>a</sup> (0.000)	0.038 <sup>a</sup> (0.000)	0.118 <sup>a</sup> (0.000)	0.057 <sup>a</sup> (0.000)	0.053 <sup>a</sup> (0.000)	0.020 <sup>a</sup> (0.001)	0.090 <sup>a</sup> (0.000)	$7.04 \times 10^{-3 a}$ (0.002)	0.100 <sup>a</sup> (0.000)	$-1.84 \times 10^{-3 a}$ (0.000)	0.125 <sup>a</sup> (0.000)	0.041 <sup>a</sup> (0.000)	0.037 <sup>a</sup> (0.000)	0.035 <sup>a</sup> (0.000)	0.080 <sup>a</sup> (0.000)	$8.93 \times 10^{-3}$ (0.315)
$\left(e_{t-1}^{u}\right)^{2}$	$8.17 \times 10^{-3 a}$ (0.001)	$-5.68 \times 10^{-3}$ (0.401)	$2.30 \times 10^{-3}$ (0.538)	-0.011 <sup>c</sup> (0.052)	$6.81 \times 10^{-3 b}$ (0.032)	-0.011 (0.108)	$4.97 \times 10^{-3}$ (0.524)	-0.011 <sup>c</sup> (0.073)	$5.98 \times 10^{-3 a}$ (0.000)	$-6.98 \times 10^{-3 \text{ c}}$ (0.083)	$-4.96 \times 10^{-3 a}$ (0.000)	-0.014 <sup>a</sup> (0.000)	$6.71 \times 10^{-3}$ (0.110)	$-8.59 \times 10^{-3}$ (0.228)	$7.89 \times 10^{-3 \text{ b}}$ (0.012)	$-9.45 \times 10^{-3}$ (0.119)
$h^o_{t-1}$	0.877 <sup>a</sup> (0.000)	-0.030 <sup>a</sup> (0.000)	0.579 <sup>a</sup> (0.000)	0.085 (0.130)	0.906 <sup>a</sup> (0.000)	-0.018 (0.290)	0.853 <sup>a</sup> (0.000)	$2.77 \times 10^{-3}$ (0.378)	0.791 <sup>a</sup> (0.000)	$1.71 \times 10^{-4}$ (0.369)	0.506 <sup>a</sup> (0.000)	-0.029 (0.142)	0.925 <sup>a</sup> (0.000)	-0.030 <sup>a</sup> (0.000)	0.813 <sup>a</sup> (0.000)	$5.47 \times 10^{-3}$ (0.599)
$h_{t-1}^u$	$-4.98 \times 10^{-3 b}$ (0.031)	0.896 <sup>a</sup> (0.000)	$6.82 \times 10^{-4}$ (0.915)	0.908 <sup>a</sup> (0.000)	$-2.59 \times 10^{-3}$ (0.407)	0.900 <sup>a</sup> (0.000)	0.010 (0.355)	0.894 <sup>a</sup> (0.000)	$-7.80 \times 10^{-3 a}$ (0.000)	0.892 <sup>a</sup> (0.000)	0.036 <sup>a</sup> (0.000)	0.903 <sup>a</sup> (0.000)	$2.08 \times 10^{-4}$ (0.968)	0.897 <sup>a</sup> (0.000)	$-4.67 \times 10^{-3 \text{ c}}$ (0.076)	0.900 <sup>a</sup> (0.000)
Asymmetry	0.098 <sup>a</sup> (0.000)	0.172 <sup>a</sup> (0.000)	0.199 <sup>a</sup> (0.000)	0.164 <sup>a</sup> (0.000)	0.068 <sup>a</sup> (0.000)	0.173 <sup>a</sup> (0.000)	0.060 <sup>a</sup> (0.000)	0.183 <sup>a</sup> (0.000)	0.143 <sup>a</sup> (0.000)	0.193 <sup>a</sup> (0.000)	0.143 <sup>a</sup> (0.000)	0.167 <sup>a</sup> (0.000)	0.051 <sup>a</sup> (0.000)	0.167 <sup>a</sup> (0.000)	0.157 <sup>a</sup> (0.000)	0.175 <sup>a</sup> (0.000)
						Par	el C: Constan	t Condition	al Correlatio	on						
<i>p</i> <sup>0,<i>u</i></sup>	0.198 <sup>a</sup> (0.000)		0.104 <sup>a</sup> (0.000)		0.185 <sup>a</sup> (0.000)		0.082 <sup>a</sup> (0.000)		0.031 <sup>b</sup> (0.030)		0.054 <sup>a</sup> (0.000)		0.142 <sup>a</sup> (0.000)		0.134 <sup>a</sup> (0.000)	
							Panel D:	Diagnostic	Tests							
LogL	30500.4		30678.873		30692.048		33342.351		30597.4		30633.0		30931.2		30683.3	
AIC SIC	-9.520		-9.912 -9.518		-10.060 -9.671		-11.074 -10.680		-10.044		-10.537		-10.030 -9.780		-10.174 -9.779	
510	-9.126		-9.518		-9.6/1		-10.660		-9.040		-10.143		-9.780		-9.//9	

	IND	USA	INDO	USA	KOR	USA	MYS	USA	PAK	USA	PHL	USA	TAIW	USA	THA	USA
							Panel D	: Diagnostic	Tests							
JB	533.870 <sup>a</sup>	381.111 <sup>a</sup>	688.360 <sup>a</sup>	528.060 <sup>a</sup>	870.230 <sup>a</sup>	391.040 <sup>a</sup>	1921.200 <sup>a</sup>	500.410 <sup>a</sup>	3442.14 <sup>a</sup>	508.010 <sup>a</sup>	10842.7 <sup>a</sup>	438.390 <sup>a</sup>	295.380 <sup>a</sup>	462.680 <sup>a</sup>	20331.6 <sup>a</sup>	560.990 <sup>a</sup>
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Q (12)	16.143	22.788	9.012	15.021	5.829	18.049	13.179	14.149	55.870 <sup>c</sup>	15.244	21.845 <sup>b</sup>	15.008	18.533 <sup>c</sup>	19.566 <sup>c</sup>	33.450 <sup>c</sup>	16.279
	(0.185)	(0.303)	(0.702)	(0.240)	(0.924)	(0.114)	(0.356)	(0.291)	(0.097)	(0.228)	(0.039)	(0.241)	(0.100)	(0.076)	(0.087)	(0.179)
Q <sup>2</sup> (12)	13.026	9.413	19.566 <sup>c</sup>	8.416	5.242	9.919	14.799	10.027	5.970	0.000 <sup>a</sup>	1.823	0.000 <sup>a</sup>	16.456	10.976	0.928	11.617
	(0.367)	(0.667)	(0.076)	(0.752)	(0.949)	(0.623)	(0.253)	(0.614)	(0.918)	(0.000)	(0.923)	(0.000)	(0.171)	(0.531)	(0.965)	(0.477)

Table 2. Cont.

Notes: The number of lags for VAR was decided using SIC (Schwartz information criterion) and AIC (Akaike information criterion) criteria. JB, Q(12), and  $Q^2(12)$  indicate the empirical statistics of the Jarque–Bera test for normality, while Ljung–Box Q statistics with order 12 for autocorrelation were applied to the standardized residuals and squared standardized residuals, respectively. USA, United States of America; IND, India; INDO, Indonesia; KOR, South Korea; MYS, Malaysia; PAK, Pakistan; PHL, the Philippines; TAIW, Taiwan; THA, Thailand. Values in parentheses are the *p*-values. <sup>a, b, c</sup> indicate the statistical significance at 1%, 5%, and 10%, respectively.

ARCH coefficient captures the shock dependence, while the GARCH coefficient captures the persistence of volatility in conditional variance equations. The findings reveal that the sensitivity of past own shocks (ARCH term) is significantly positive for all Asian Stock Markets in the short run. In addition, the sensitivity of past own volatility (the GARCH term) was found to be significant for all stock markets (including the Asian and American Markets), thus the ARCH (1) volatility model was determined to be more appropriate in this case. The coefficient of past own volatility was than the coefficients of past own shocks in all Asian stock markets, implying that past own volatilities are more critical for prediction of future volatility as compared to past own shocks.

The conditional volatility of India's, South Korea's, the Philippines', Pakistan's, and Thailand's stock markets was found to be significantly affected by shocks in the American stock market. These results are similar to the findings of Syriopoulos et al. (2015), which show that past shocks in the American market significantly affect the market volatility of India, Brazil, and Russia. Therefore, this implies that shock in the American stock market leads to an increase in the volatility of the majority of Asian markets. The past volatility of the American stock market significantly influenced the conditional volatility of India's, The Philippines', Pakistan's, and Thailand's stock markets. These results confirm the previous findings of Li and Giles (2015), which finds a significant volatility spillover from the USA to emerging Asian stock markets. Further, Syriopoulos et al. 2015 found a significant volatility spillover from the USA to India. In addition, the past volatility of the American stock market. The asymmetric coefficients of all Asian stock markets are significant and positive, showing that negative news (or unexpected shocks) for the American stock market has more ability to increase the volatility of all Asian Stock markets as compared to positive news.

Besides, the asymmetric coefficient of the American stock market is positively significant, demonstrating that negative unexpected shocks in Asian Stock markets will increase the volatility more in the American Stock market as compared to a positive shock. Constant conditional correlation (CCC) is positively significant for all pairs of stock markets. However, cross-market correlation is weak in almost all pairs, indicating that investors can get substantial gains by having these pairs in the same portfolio.

#### 4.2.2. Stock Market Linkages between China and Asia from the Full Sample Period

Table 3 reports the return and volatility spillover between the Chinese and other Asian stock markets during the full sample period. The current stock returns of Asian stock markets are significantly affected by their own lagged stock returns. This highlights the possibility of short-term predictions of current returns through past returns in the Asian stock markets. Moreover, Chinese stock returns are also significantly influenced by their own single period lagged returns. These findings depict that stock prices can be predicted in the short term in the Chinese stock market.

The return spillover is not significant from China to the majority of other Asian markets except for the Indian, Philippines, and Thai stock markets. Besides, the return transmission from Asian markets to the Chinese market is insignificant except for in the case of the Indian Stock market. Moreover, there is a presence of bi-directional return transmission between the Indian and Chinese stock markets. This implies that Chinese (Indian) stock market prices play an important role in predicting the prices of Indian (Chinese) stock markets during the full sample period. The coefficient of past own shock of all Asian markets (including China) was found to be significant; thus, past shocks affect current conditional volatility in Asian stock markets. Besides, the sensitivity of past own volatility for all Asian markets was found to be significant as well.

	IND	CHN	INDO	CHN	KOR	CHN	MYS	CHN	РАК	CHN	PHL	CHN	TAIW	CHN	THA	CHN
							Panel A:	: Mean Equ	ation							
Constant	$5.06 \times 10^{-4 a}$ (0.000)	$3.25 \times 10^{-4 \text{ c}}$ (0.053)	$5.37 \times 10^{-4 a}$ (0.000)	$3.50 \times 10^{-4 \text{ b}}$ (0.022)	$2.71 \times 10^{-4 \text{ c}}$ (0.065)	$3.26 \times 10^{-4 \text{ b}}$ (0.043)	$1.69 \times 10^{-4 \text{ b}}$ (0.036)	$3.74 \times 10^{-4 \text{ b}}$ (0.023)	$8.64 \times 10^{-4 a}$ (0.000)	$3.39 \times 10^{-4 \text{ b}}$ (0.029)	$3.32 \times 10^{-4 \text{ b}}$ (0.031)	$3.50 \times 10^{-4 \text{ b}}$ (0.030)	$2.18 \times 10^{-4 \text{ c}}$ (0.089)	$3.51 \times 10^{-4 b}$ (0.027)	$5.05 \times 10^{-4 a}$ (0.000)	$2.12 \times 10^{-4}$ (0.139)
$r_{t-1}^o$	0.137 <sup>a</sup> (0.000)	0.036 <sup>a</sup> (0.004)	0.152 <sup>a</sup> (0.000)	0.017 (0.237)	0.083 <sup>a</sup> (0.000)	0.011 (0.345)	0.191 <sup>a</sup> (0.000)	0.018 (0.456)	0.167 <sup>a</sup> (0.000)	$6.27 \times 10^{-3}$ (0.576)	0.144 <sup>a</sup> (0.000)	-0.028 <sup>b</sup> (0.048)	0.100 <sup>a</sup> (0.000)	0.017 (0.241)	0.125 <sup>a</sup> (0.000)	0.018 (0.183)
$r_{t-1}^c$	-0.018 <sup>c</sup> (0.096)	0.049 <sup>a</sup> (0.001)	$-6.05 \times 10^{-3}$ (0.603)	0.050 <sup>a</sup> (0.000)	-0.013 (0.269)	0.055 <sup>a</sup> (0.000)	7.11 $\times$ 10 <sup>-3</sup> (0.295)	0.053 <sup>a</sup> (0.000)	0.010 (0.272)	0.055 <sup>a</sup> (0.000)	0.025 <sup>b</sup> (0.043)	0.055 <sup>a</sup> (0.000)	$2.26 \times 10^{-3}$ (0.841)	0.051 <sup>a</sup> (0.000)	-0.020 (0.103)	0.049 <sup>a</sup> (0.001)
							Panel B: V	Variance Eq	uation							
Constant	2.14 × 10-6 <sup>a</sup> (0.000)	1.14 × 10-6 <sup>a</sup> (0.001)	4.30 × 10-6 <sup>a</sup> (0.000)	1.19 × 10-6 <sup>a</sup> (0.002)	1.00 × 10-6 <sup>a</sup> (0.000)	1.37 × 10-6 <sup>a</sup> (0.000)	5.84 × 10-7 <sup>a</sup> (0.000)	1.26 × 10-6 <sup>a</sup> (0.000)	7.29 × 10-6 <sup>a</sup> (0.000)	1.15 × 10-6 <sup>a</sup> (0.002)	1.36 × 10-5 <sup>a</sup> (0.000)	1.19 × 10-6 <sup>b</sup> (0.048)	9.17 × 10-7 <sup>a</sup> (0.000)	1.21 × 10-6 <sup>a</sup> (0.000)	1.01 × 10-6 <sup>a</sup> (0.001)	1.19 × 10-6 <sup>a</sup> (0.001)
$\left(e_{t-1}^{o}\right)^{2}$	0.051 <sup>a</sup> (0.000)	$-1.95 \times 10^{-3}$ (0.413)	0.074 <sup>a</sup> (0.000)	$-1.77 \times 10^{-4}$ (0.950)	0.031 <sup>a</sup> (0.000)	$7.03 \times 10^{-3}$ (0.241)	0.074 <sup>a</sup> (0.000)	$3.80 \times 10^{-4}$ (0.668)	0.100 <sup>a</sup> (0.000)	$-9.47 \times 10^{-4}$ (0.509)	0.053 <sup>a</sup> (0.000)	$8.20 \times 10^{-3 b}$ (0.016)	0.033 <sup>a</sup> (0.000)	$3.74 \times 10^{-3 b}$ (0.034)	0.070 <sup>a</sup> (0.000)	0.023 <sup>a</sup> (0.000)
$\left(e_{t-1}^{c}\right)^{2}$	-5.03 × 10 <sup>-3 c</sup> (0.081)	0.069 <sup>a</sup> (0.000)	9.29 × 10 <sup>-3 c</sup> (0.099)	0.066 <sup>a</sup> (0.000)	$-1.32 \times 10^{-3}$ (0.551)	0.070 <sup>a</sup> (0.000)	$-6.92 \times 10^{-3}$ (0.422)	0.069 <sup>a</sup> (0.000)	$3.56 \times 10^{-3}$ (0.453)	0.064 <sup>a</sup> (0.000)	$8.77 \times 10^{-4}$ (0.754)	0.068 <sup>a</sup> (0.000)	9.94 × 10 <sup>-3 a</sup> (0.000)	0.069 <sup>a</sup> (0.000)	$7.51 \times 10^{-3 b}$ (0.025)	0.062 <sup>a</sup> (0.000)
$h^o_{t-1}$	0.876 <sup>a</sup> (0.000)	$9.48 \times 10^{-3 a}$ (0.001)	0.840 <sup>a</sup> (0.000)	0.011 <sup>b</sup> (0.041)	0.929 <sup>a</sup> (0.000)	$-6.06 \times 10^{-3 a}$ (0.000)	0.871 <sup>a</sup> (0.000)	2.91 × 10 <sup>-3 b</sup> (0.022)	0.792 <sup>a</sup> (0.000)	$1.39 \times 10^{-3}$ (0.458)	0.797 <sup>a</sup> (0.000)	$2.22 \times 10^{-3}$ (0.656)	0.922 <sup>a</sup> (0.000)	$-1.34 \times 10^{-3}$ (0.508)	0.878 <sup>a</sup> (0.000)	$-8.97 \times 10^{-3 a}$ (0.001)
$h_{t-1}^c$	$7.61 \times 10^{-3 b}$ (0.033)	0.920 <sup>a</sup> (0.000)	$-4.86 \times 10^{-3}$ (0.482)	0.923 <sup>a</sup> (0.000)	$2.24 \times 10^{-3}$ (0.314)	0.918 <sup>a</sup> (0.000)	0.010 (0.278)	0.921 <sup>a</sup> (0.000)	$-1.29 \times 10^{-3}$ (0.816)	0.923 <sup>a</sup> (0.000)	$1.53 \times 10^{-3}$ (0.816)	0.921 <sup>a</sup> (0.000)	$-8.00 \times 10^{-3 a}$ (0.002)	0.921 <sup>a</sup> (0.000)	$-3.06 \times 10^{-3}$ (0.497)	0.919 <sup>a</sup> (0.000)
Asymmetry	0.105 <sup>a</sup> (0.000)	0.016 <sup>b</sup> (0.048)	0.096 <sup>a</sup> (0.000)	0.012 (0.138)	0.067 <sup>a</sup> (0.000)	0.019 <sup>b</sup> (0.023)	0.073 <sup>a</sup> (0.000)	0.016 <sup>c</sup> (0.055)	0.139 <sup>a</sup> (0.000)	0.019 <sup>b</sup> (0.012)	0.096 <sup>a</sup> (0.000)	0.018 <sup>b</sup> (0.035)	0.072 <sup>a</sup> (0.000)	0.016 <sup>b</sup> (0.050)	0.073 <sup>a</sup> (0.000)	0.030 <sup>a</sup> (0.001)
						Pa	nel C: Constan	t Condition	al Correlati	on						
$p^{0,c}$	0.158 <sup>a</sup> (0.000)		0.164 <sup>a</sup> (0.000)		0.211 <sup>a</sup> (0.000)		a (0.000)		0.047 <sup>a</sup> (0.001)		0.131 <sup>a</sup> (0.000)		0.218 <sup>a</sup> (0.000)		0.146 <sup>a</sup> (0.000)	
							Panel D:	Diagnostic	Tests							
LogL	28559.5		28690.5		28584.3		31339.5		28776.2		28564.8		28882.9		28821.945	
AIC	9.191		-9.543		-9.621		-10.762		-9.824		-9.920		-9.766		-9.770	
SIC	-8.797		-9.149		-9.227		-10.368		-9.430		-9.520		-9.372		-9.375	

**Table 3.** Estimates of bivariate VAR–AGARCH for China's and other Asian stock markets during the full sample period.

	IND	CHN	INDO	CHN	KOR	CHN	MYS	CHN	PAK	CHN	PHL	CHN	TAIW	CHN	THA	CHN
							Panel D	: Diagnostic	Tests							
JB	574.750 <sup>a</sup>	1348.92 <sup>a</sup>	1310.07 <sup>a</sup>	1391.71 <sup>a</sup>	1385.22 <sup>a</sup>	705.970 <sup>a</sup>	1651.18 <sup>a</sup>	1428.96 <sup>a</sup>	3079.52 <sup>a</sup>	1424.88 <sup>a</sup>	1248.20 <sup>a</sup>	1408.79 <sup>a</sup>	464.070 <sup>a</sup>	1238.55 <sup>a</sup>	56026.9 <sup>a</sup>	1380.10 <sup>a</sup>
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Q (12)	17.399	48.803 <sup>a</sup>	10.709	45.754 <sup>a</sup>	9.252	46.376 <sup>a</sup>	15.229	44.283 <sup>a</sup>	54.789 <sup>a</sup>	51.513 <sup>a</sup>	14.166	50.473 <sup>a</sup>	24.449 <sup>b</sup>	39.022 <sup>a</sup>	31.740 <sup>a</sup>	48.687 <sup>a</sup>
	(0.135)	(0.000)	(0.554)	(0.000)	(0.681)	(0.000)	(0.229)	(0.000)	(0.000)	(0.000)	(0.290)	(0.000)	(0.018)	(0.000)	(0.002)	(0.000)
$Q^{2}(12)$	11.249	11.647	17.033	9.322	5.203	11.672	20.075 <sup>c</sup>	11.915	6.182	12.065	4.715	10.218	11.938	13.649	2.396	10.561
	(0.508)	(0.474)	(0.148)	(0.675)	(0.951)	(0.472)	(0.066)	(0.453)	(0.907)	(0.440)	(0.967)	(0.597)	(0.451)	(0.324)	(0.999)	(0.567)

Table 3. Cont.

Notes: The number of lags for VAR was decided using SIC (Schwartz information criterion) and AIC (Akaike information criterion) criteria. JB, Q(12) and Q<sup>2</sup>(12) indicated the empirical statistics of the Jarque–Bera test for normality, while Ljung–Box Q statistics of order 12 for autocorrelation applied to the standardized residuals and squared standardized residuals, respectively. CHN, China; IND, India; INDO, Indonesia; KOR, South Korea; MYS, Malaysia; PAK, Pakistan; PHL, the Philippines; TAIW, Taiwan; THA, Thailand. Values in parentheses are the *p*-values. <sup>a</sup>, <sup>b</sup>, <sup>c</sup></sup> indicate the statistical significance at 1%, 5%, and 10%, respectively.

The conditional volatility of India, Indonesia, Taiwan, and Thailand is significantly affected by shocks in the Chinese market. Also, the conditional volatility of the Chinese market is significantly impacted by the shocks in the Philippines, Taiwanese, and Thai stock markets. The past volatility of the Chinese stock market has not influenced the conditional volatility of the most of the Asian stock markets except for the Indian and Taiwanese stock markets. These findings corroborate with the results of Zhou et al. (2012), which report a significant spillover from China to the Taiwanese stock market. However, the past volatility of the majority Asian markets (except for Pakistan, the Philippines, and Taiwan) significantly affected the conditional volatility of the Chinese stock market.

The asymmetric coefficients of all Asian stock markets were found to be significant and positive, showing that negative news of the Chinese stock market has more of an ability to increase the volatility of all Asian stock markets as compared to positive news. Moreover, the asymmetric coefficient of the Chinese stock market is significant and positive, showing that negative news in Asian markets (except in Indonesia) has a greater ability to increase the volatility of the Chinese market as compared to positive news. Constant conditional correlation is positively significant for all pairs of stock markets, but CCC is weak in majority pairs.

#### 4.2.3. Stock Market Linkages between the USA and Asia from the US Financial Crisis

Table 4 shows the mean and volatility spillover between the USA and Asian stock markets during the US financial crisis. In Asian Stock markets (except for South Korea), past lagged returns significantly influenced the current returns. This highlights the possibility of short-term prediction of current returns through past returns in the Asian stock markets. Moreover, the American stock returns were also significantly influenced by their own single period lagged returns in the majority of cases.

The return spillover effect from the USA to all Asian markets was seen to be significant during the US financial crisis. This implies that US stock market prices played an important role in predicting the prices of all Asian stock markets during the US financial crisis. These results confirm the previous findings of Glick and Hutchison (2013), who reported a significant impact of American equity returns on Asian equity returns during the US financial crisis. Moreover, no single Asian stock market transmitted the return effect to the American market during the US financial crisis. The sensitivity of past own shock was significant for the majority of Asian markets other than Indonesia, Korea, and Taiwan. The coefficient of past own shocks of the American stock market was insignificant in the majority estimations. Besides, the coefficient of own past volatility in all Asian markets was significant except in the Philippines.

The past shocks in the American stock market significantly influenced the conditional volatility of Korea, the Philippines, and Taiwan during the US financial crisis. However, past shocks in most of the Asian stock markets (Except India) have not affected the conditional volatility of the American stock market. The effect of past volatility in the USA on conditional volatility of the Asian stock markets (except Korea) was found to be insignificant. These results match with the findings of Li and Giles (2015), which observe an absence of volatility spillover from the USA to emerging Asian stock markets during the US financial crisis. Moreover, the past volatility of majority Asian stock has not significantly affected American stock market volatility. The asymmetric coefficient of all Asian markets is significant and positive. Moreover, the asymmetric coefficient of the US market is significant and positive in all cases. Constant conditional correlation is positively significant for all pairs of stock markets, but CCC is weak in majority pairs.

	IND	USA	INDO	USA	KOR	USA	MYS	USA	PAK	USA	PHL	USA	TAIW	USA	THA	USA
							Panel A	: Mean Equa	ation							
Constant	$7.60 \times 10^{-4}$ (0138)	$3.49 \times 10^{-4}$ (0.936)	$-2.0 \times 10^{-5}$ (0957)	$8.03 \times 10^{-5}$ (0.834)	$8.5 \times 10^{-4 \text{ c}}$ (0.080)	$-7.01 \times 10^{-5}$ (0.865)	$3.72 \times 10^{-4}$ (0.123)	$-1.43 \times 10^{-4}$ (0.742)	$9.53 \times 10^{-4 a}$ (0.005)	$3.57 \times 10^{-4}$ (0.262)	$9.83 \times 10^{-4 a}$ (0.001)	$4.79 \times 10^{-4 a}$ (0.066)	$1.90 \times 10^{-4}$ (0.684)	$-2.64 \times 10^{-5}$ (0.955)	$6.00 \times 10^{-4}$ (0.194)	$-6.12 \times 10^{-5}$ (0.886)
$r^o_{t-1}$	0.081 <sup>b</sup> (0.031)	0.016 (0.510)	0.031 <sup>b</sup> (0.036)	-0.024 (0.445)	0.044 (0.141)	-0.010 (0.713)	0.145 <sup>a</sup> (0.000)	-0.046 (0.409)	0.125 <sup>a</sup> (0.001)	0.015 (0.648)	0.101 <sup>a</sup> (0.000)	0.016 (0.523)	0.063 <sup>c</sup> (0.076)	0.020 (0.565)	0.065 <sup>b</sup> (0.027)	-0.012 (0.663)
$r_{t-1}^u$	0.293 <sup>a</sup> (0.000)	-0.093 <sup>b</sup> (0.016)	0.423 <sup>a</sup> (0.000)	-0.062 <sup>a</sup> (0.091)	0.325 <sup>a</sup> (0.000)	-0.079 <sup>b</sup> (0.041)	0.187 <sup>a</sup> (0.000)	-0.080 <sup>b</sup> (0.042)	0.175 <sup>a</sup> (0.000)	0.016 (0.706)	0.489 <sup>a</sup> (0.000)	$1.60 \times 10^{-3}$ (0.963)	0.383 <sup>a</sup> (0.000)	-0.103 <sup>b</sup> (0.011)	0.249 <sup>a</sup> (0.000)	-0.077 <sup>b</sup> (0.034)
							Panel B: V	Variance Equ	uation							
Constant	$2.91 \times 10^{-6}$ (0.161)	$3.39 \times 10^{-6 a}$ (0.001)	$1.33 \times 10^{-5 a}$ (0.000)	$5.10 \times 10^{-6 a}$ (0.000)	$3.98 \times 10^{-5 a}$ (0.000)	$4.34 \times 10^{-6 a}$ (0.000)	$1.80 \times 10^{-6 \text{ b}}$ (0.038)	$3.40 \times 10^{-6 a}$ (0.001)	$4.22 \times 10^{-6 a}$ (0.007)	$3.93 \times 10^{-6 a}$ (0.000)	$3.87 \times 10^{-5 a}$ (0.000)	$2.98 \times 10^{-6 a}$ (0.000)	$1.15 \times 10^{-6}$ (0.352)	$3.73 \times 10^{-6 a}$ (0.009)	$3.72 \times 10^{-5 a}$ (0.000)	$4.59 \times 10^{-8}$ (0.984)
$\left(e_{t-1}^{o}\right)^{2}$	0.086 <sup>a</sup> (0.000)	0.075 <sup>b</sup> (0.016)	0.198 <sup>a</sup> (0.000)	0.001 (0.877)	0.012 <sup>a</sup> (0.000)	0.077 <sup>c</sup> (0.078)	0.119 <sup>a</sup> (0.000)	$-7.24 \times 10^{-4}$ (0.879)	0.036 <sup>b</sup> (0.034)	$7.53 \times 10^{-3}$ (0.439)	0.283 (0.140)	0.051 <sup>b</sup> (0.044)	$4.98 \times 10^{-3}$ (0.761)	0.029 (0.230)	0.151 <sup>a</sup> (0.000)	0.015 (0.678)
$\left(e_{t-1}^{u}\right)^{2}$	$6.65 \times 10^{-3 a}$ (0.053)	$-0.01 \times 10^{-3}$ (0.337)	0.011 (0.138)	-0.029 <sup>b</sup> (0.014)	$7.71 \times 10^{-3 b}$ (0.052)	0.001 (0.917)	0.051 (0.213)	-0.013 (0.361)	0.018 (0.146)	-0.060 <sup>a</sup> (0.001)	$-4.01 \times 10^{-3 \text{ b}}$ (0.033)	-0.061 <sup>a</sup> (0.000)	0.026 <sup>c</sup> (0.056)	-0.022 (0.113)	$-5.15 \times 10^{-3}$ (0.690)	$-6.17 \times 10^{-3}$ (0.661)
$h^o_{t-1}$	-0.868 <sup>a</sup> (0.000)	-0.048 <sup>b</sup> (0.012)	0.750 <sup>a</sup> (0.000)	0.060 (0.208)	0.426 <sup>a</sup> (0.000)	-0.012 (0.330)	0.755 <sup>a</sup> (0.000)	0.013 <sup>c</sup> (0.090)	0.854 <sup>a</sup> (0.000)	$8.26 \times 10^{-3}$ (0.470)	0.099 (0.196)	0.043 (0.171)	0.951 <sup>a</sup> (0.000)	-0.029 <sup>c</sup> (0.062)	0.449 <sup>a</sup> (0.000)	0.110 <sup>b</sup> (0.024)
$h_{t-1}^u$	$2.17 \times 10^{-3 c}$ (0.087)	0.894 <sup>a</sup> (0.000)	$5.62 \times 10^{-4}$ (0.765)	0.945 <sup>a</sup> (0.000)	-0.176 (0.394)	0.906 <sup>a</sup> (0.000)	-0.035 (0.312)	0.915 <sup>a</sup> (0.000)	-0.021 (0.322)	0.907 <sup>a</sup> (0.000)	$8.49 \times 10^{-3}$ (0.115)	0.906 <sup>a</sup> (0.000)	-0.017 (0.382)	0.910 <sup>a</sup> (0.000)	0.031 (0.299)	0.905 <sup>a</sup> (0.000)
Asymmetry	0.064 <sup>c</sup> (0.090)	0.179 <sup>a</sup> (0.000)	0.265 <sup>a</sup> (0.000)	0.159 <sup>a</sup> (0.000)	0.446 <sup>a</sup> (0.000)	0.161 <sup>a</sup> (0.000)	0.163 <sup>a</sup> (0.004)	0.154 <sup>a</sup> (0.000)	0.101 <sup>a</sup> (0.004)	0.237 <sup>a</sup> (0.000)	0.523 <sup>a</sup> (0.000)	0.233 <sup>a</sup> (0.000)	0.075 <sup>a</sup> (0.000)	0.169 <sup>a</sup> (0.000)	0.187 <sup>a</sup> (0.000)	0.147 <sup>a</sup> (0.000)
						Pai	nel C: Constan	t Condition	al Correlatio	on						
<i>p</i> <sup>0,<i>u</i></sup>	0.172 <sup>a</sup> (0.000)		0.280 <sup>a</sup> (0.000)		0.220 <sup>a</sup> (0.000)		0.195 <sup>a</sup> (0.000)		0.016 <sup>c</sup> (0.086)		0.065 <sup>b</sup> (0.044)		0.187 <sup>a</sup> (0.000)		0.218 <sup>a</sup> (0.000)	
							Panel D:	Diagnostic	Tests							
LogL	4230.27		4320.18		4436.36		4843.96		4850.88		6628.56		4391.36		4399.67	
AIC	-10.066		-10.502		-10.394		-11.480		-11.499		-11.518		-10.601		-10.710	
SIC	-9.773		-10.209		-10.101		-11.187		-11.205		-11.224		-10.308		-10.417	

Table 4. Estimates of bivariate VAR–AGARCH for the American and Asian stock markets during the US Financial Crisis.

	IND	USA	INDO	USA	KOR	USA	MYS	USA	PAK	USA	PHL	USA	TAIW	USA	THA	USA
							Panel D:	Diagnostic	Tests							
JB	513.31 <sup>a</sup>	372.23 <sup>a</sup>	679.01 <sup>a</sup>	512.65 <sup>a</sup>	843.871 <sup>a</sup>	399.862 <sup>a</sup>	321.220 <sup>a</sup>	627.410	676.080 <sup>a</sup>	654.370 <sup>a</sup>	1598.28 <sup>a</sup>	629.470 <sup>a</sup>	352.580 <sup>a</sup>	635.280 <sup>a</sup>	330.040 <sup>a</sup>	643.320 <sup>a</sup>
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Q (12)	14.588	8.939	11.706	6.002	12.006	7.401	11.897	6.280	17.579	9.780	9.776	7.878	14.047	5.797	13.234	5.468
	(0.264)	(0.708)	(0.469)	(0.915)	(0.445)	(0.829)	(0.454)	(0.901)	(0.129)	(0.635)	(0.636)	(0.795)	(0.298)	(0.926)	(0.352)	(0.941)
Q <sup>2</sup> (12)	7.779	13.979	5.303	14.983	6.336	13.673	5.422	14.540	6.417	32.448 <sup>a</sup>	2.113	20.023 <sup>c</sup>	6.158	13.940	8.596	14.294
	(0.802)	(0.301)	(0.947)	(0.242)	(0.898)	(0.322)	(0.942)	(0.268)	(0.894)	(0.001)	(0.999)	(0.067)	(0.908)	(0.305)	(0.737)	(0.282)

Table 4. Cont.

Notes: The number of lags for VAR was decided using SIC (Schwartz information criterion) and AIC (Akaike information criterion) criteria. JB, Q(12) and  $Q^2(12)$  indicate the empirical statistics of the Jarque–Bera test for normality, while Ljung–Box Q statistics of order 12 for autocorrelation were applied to the standardized residuals and squared standardized residuals, respectively. USA, United States of America; IND, India; INDO, Indonesia; KOR, South Korea; MYS, Malaysia; PAK, Pakistan; PHL, the Philippines; TAIW, Taiwan; THA, Thailand. Values in parentheses are the *p*-values. <sup>a</sup>, <sup>b</sup>, <sup>c</sup> indicate the statistical significance at 1%, 5%, and 10%, respectively.

Table 5 reports the return and volatility spillover between China and Asian stock markets during the US financial crisis. The current stock returns of the majority of Asian stock markets (Except in South Korea) are significantly affected by their own lagged stock returns. This highlights the possibility of short-term prediction of current returns through past returns in the Asian stock markets. However, Chinese stock returns were not significantly affected by their lagged returns during the US financial crisis. Therefore, there is no evidence of Chinese stock price prediction being possible through lagged values during the US financial crisis.

The return transmission effect from China to all Asian markets was insignificant during the US financial US crisis. However, most of the Asian markets did not transmit the return effect to the Chinese stock market other than India, Indonesia, and Malaysia. The coefficient of past own shock was found to be significant in the majority of Asian markets except for Indonesia, Korea, and Pakistan. The sensitivity to past own shocks from Chinese stock markets was found to be insignificant in the majority of markets during the US financial crisis. Moreover, the sensitivity of past own volatility in all Asian markets was significant. The past shocks of China did not influence the conditional volatility of the majority of Asian stock markets (except India) during the US financial crisis. The conditional volatility of the Chinese stock market was not affected by shocks in most of the Asian stock markets (except for Indonesia and Thailand).

There is no significant evidence of volatility spillover from Chinese to Asian stock markets except in India and Taiwan. Besides, the volatility spillover was insignificant in the majority of Asian markets (except Indonesia, Pakistan, and The Philippines) to the Chinese stock market. The asymmetric coefficient of all Asian markets is significant and positive. Moreover, the asymmetric coefficient of China is asymmetric, showing that that negative news of all Asian stock markets (except Pakistan) has more ability to increase the volatility of the Chinese stock market as compared to positive news. Constant conditional correlation is positively significant for all pairs of stock markets. However, CCC has a medium level in the majority of pairs.

#### 4.2.5. Stock Market Linkages between the USA and Asia from the Chinese Stock Market Crash

Table 6 reports the mean and volatility spillover between the USA and Asian stock markets during the Chinese stock market crash. The autoregressive term of Asian market returns (Except Korea, the Philippines, and Taiwan) can be seen to be significant in the majority of stock markets. This shows the short-term predictability in stock price changes in the Asian stock markets. In addition, stock returns of the American stock market were significantly influenced by their lagged returns during the Chinese Crisis.

	IND	CHN	INDO	CHN	KOR	CHN	MYS	CHN	PAK	CHN	PHL	CHN	TAIW	CHN	THA	CHN
							Panel A:	Mean Equa	ation							
Constant	$6.04 \times 10^{-4}$	$-2.82 \times 10^{-4}$	$6.72 \times 10^{-4}$	$-2.28 \times 10^{-4}$	$-7.49 \times 10^{-5}$	$-2.18 \times 10^{-5}$	$4.34 \times 10^{-4}$	$-5.07 \times 10^{-5}$	$7.61 \times 10^{-4 b}$	$1.87 \times 10^{-4}$	$2.57 \times 10^{-4}$	$-8.84 \times 10^{-5}$	$2.65 \times 10^{-4}$	$3.46 \times 10^{-5}$	$7.98 \times 10^{-4 \text{ c}}$	$9.09 \times 10^{-5}$
	(0.2680	(0.671)	(0.220)	(0.712)	(0.867)	(0.974)	(0.146)	(0.943)	(0.017)	(0.684)	(0.615)	(0.895)	(0.580)	(0.954)	(0.091)	(0.876)
$r_{t-1}^o$	0.147 <sup>a</sup>	0.109 <sup>a</sup>	0.146 <sup>a</sup>	0.136 <sup>a</sup>	0.052	$8.15 \times 10^{-3}$	0.216 <sup>a</sup>	0.196 <sup>a</sup>	0.117 <sup>a</sup>	-0.032	0.138 <sup>a</sup>	$5.75 \times 10^{-4}$	0.102 <sup>a</sup>	0.017	0.128 <sup>a</sup>	0.045
	(0.000)	(0.002)	(0.000)	(0.001)	(0.211)	(0.861)	(0.000)	(0.008)	(0.000)	(0.451)	(0.001)	(0.990)	(0.007)	(0.689)	(0.000)	(0.268)
$r_{t-1}^c$	-0.044	0.022	$-9.00 \times 10^{-3}$	0.019	0.011	0.067 <sup>c</sup>	9.47 × $10^{-3}$	0.030	0.029	0.031	$-7.18 \times 10^{-3}$	0.043	0.015	0.052	-0.025	0.054
	(0.135)	(0.576)	(0.728)	(0.614)	(0.675)	(0.086)	(0.534)	(0.432)	(0.259)	(0.432)	(0.773)	(0.282)	(0.570)	(0.183)	(0.233)	(0.147)
-							Panel B: V	/ariance Equ	uation							
Constant	$5.43 \times 10^{-6 \text{ c}}$	$1.43 \times 10^{-5 a}$	$3.74 \times 10^{-5 a}$	$3.62 \times 10^{-6}$	$3.52 \times 10^{-6}$	$1.61 \times 10^{-5 a}$	$2.63 \times 10^{-6 \text{ c}}$	$1.48 \times 10^{-5 a}$	$-2.77 \times 10^{-6}$	$3.62 \times 10^{-6 \text{ c}}$	$5.69 \times 10^{-5 a}$	$1.70 \times 10^{-5 \text{ b}}$	$1.84 \times 10^{-6}$	$1.10 \times 10^{-5 a}$	$1.50 \times 10^{-5 a}$	$8.77 \times 10^{-6}$
	(0.064)	(0.002)	(0.001)	(0.389)	(0.101)	(0.004)	(0.087)	(0.004)	(0.365)	(0.064)	(0.000)	(0.043)	(0.223)	(0.050)	(0.007)	(0.144)
$\left(e_{t-1}^{o}\right)^{2}$	0.067 <sup>a</sup>	$7.41 \times 10^{-3}$	$5.31 \times 10^{-3}$	-0.025 <sup>b</sup>	-0.016	$2.75 \times 10^{-3}$	0.112 <sup>a</sup>	$-6.75 \times 10^{-4}$	-0.018	$5.44 \times 10^{-3}$	0.070 <sup>a</sup>	$1.27 \times 10^{-3}$	0.022	$7.35 \times 10^{-3}$	0.055 <sup>b</sup>	-0.011 <sup>b</sup>
	(0.001)	(0.446)	(0.821)	(0.013)	(0.271)	(0.636)	(0.000)	(0.819)	(0.269)	(0.489)	(0.001)	(0.870)	(0.108)	(0.229)	(0.019)	(0.044)
$\left(e_{t-1}^{c}\right)^{2}$	$-7.19 \times 10^{-3 b}$	$-2.74 \times 10^{-3}$	$-6.98 \times 10^{-3}$	0.027 <sup>c</sup>	0.015	$9.48 \times 10^{-3}$	0.100	0.020	$-7.46 \times 10^{-3}$	0.025 <sup>b</sup>	$1.48 \times 10^{-3}$	0.017	-0.018	0.013	$-6.48 \times 10^{-3}$	0.012
	(0.017)	(0.870)	(0.748)	(0.070)	(0.398)	(0.587)	(0.113)	(0.265)	(0.613)	(0.047)	(0.864)	(0.391)	(0.184)	(0.388)	(0.704)	(0.468)
$h^o_{t-1}$	0.893 <sup>a</sup>	-0.016	0.623 <sup>a</sup>	0.073 <sup>c</sup>	0.919 <sup>a</sup>	$-7.47 \times 10^{-4}$	0.789 <sup>a</sup>	$3.24 \times 10^{-3}$	0.844 <sup>a</sup>	0.064 <sup>a</sup>	0.726 <sup>a</sup>	-0.042 <sup>b</sup>	0.936 <sup>a</sup>	$-5.07 \times 10^{-3}$	0.834 <sup>a</sup>	$4.85 \times 10^{-3}$
	(0.000)	(0.210)	(0.000)	(0.049)	(0.000)	(0.924)	(0.000)	(0.615)	(0.000)	(0.015)	(0.000)	(0.023)	(0.000)	(0.540)	(0.000)	(0.637)
$h_{t-1}^c$	0.030 <sup>a</sup>	0.865 <sup>a</sup>	0.060	0.896 <sup>a</sup>	$1.15 \times 10^{-3}$	0.879 <sup>a</sup>	$7.69 \times 10^{-3}$	0.865 <sup>a</sup>	$2.14 \times 10^{-3}$	0.952 <sup>a</sup>	0.016	0.867 <sup>a</sup>	0.041 <sup>c</sup>	0.885 <sup>a</sup>	0.044	0.876 <sup>a</sup>
	(0.002)	(0.000)	(0.157)	(0.000)	(0.952)	(0.000)	(0.916)	(0.000)	(0.909)	(0.000)	(0.504)	(0.000)	(0.099)	(0.000)	(0.128)	(0.000)
Asymmetry	0.091 <sup>a</sup>	0.151 <sup>a</sup>	0.361 <sup>a</sup>	0.054 <sup>b</sup>	0.152 <sup>a</sup>	0.120 <sup>a</sup>	0.149 <sup>a</sup>	0.109 <sup>a</sup>	0.167 <sup>a</sup>	0.010	0.153 <sup>a</sup>	0.126 <sup>a</sup>	0.059 <sup>a</sup>	0.116 <sup>a</sup>	0.124 <sup>a</sup>	0.130 <sup>a</sup>
	(0.003)	(0.000)	(0.000)	(0.044)	(0.000)	(0.001)	(0.006)	(0.001)	(0.000)	(0.451)	(0.005)	(0.000)	(0.005)	(0.000)	(0.001)	(0.000)
						Par	nel C: Constan	t Condition	al Correlatio	on						
<i>p</i> <sup>0,c</sup>	0.307 <sup>a</sup> (0.000)		0.290 <sup>a</sup> (0.000)		0.401 <sup>a</sup> (0.000)		0.287 <sup>a</sup> (0.000)		0.103 <sup>a</sup> (0.001)		0.270 <sup>a</sup> (0.000)		0.373 <sup>a</sup> (0.000)		0.269 <sup>a</sup> (0.000)	
							Panel D:	Diagnostic	Tests							
LogL	3980.79		4085.73		4161.31		4570.93		4646.76		4104.40		4125.88		4144.48	
AIC SIC	-9.451		-9.746		-9.840		-10.871		-11.412		-9.940		-9.990		-10.001	
SIC	-9.158		-9.453		-9.547		-10.578		-11.118		-9.650		-9.770		-9.707	

Table 5. Estimates of bivariate VAR–AGARCH for China and Asian stock markets during the US F	inancial Crisis.

	IND	CHN	INDO	CHN	KOR	CHN	MYS	CHN	PAK	CHN	PHL	CHN	TAIW	CHN	THA	CHN
							Panel D:	Diagnostic	Tests							
JB	501.810 <sup>a</sup>	693.810 <sup>a</sup>	541.320 <sup>a</sup>	686.040 <sup>a</sup>	374.450 <sup>a</sup>	673.900 <sup>a</sup>	541.990 <sup>a</sup>	694.160	590.770 <sup>a</sup>	756.610 <sup>a</sup>	1903.28 <sup>a</sup>	678.760 <sup>a</sup>	356.750 <sup>a</sup>	671.310 <sup>a</sup>	358.350 <sup>a</sup>	686.450 <sup>a</sup>
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Q (12)	14.715	7.454	11.152	7.664	12.457	8.501	11.203	6.933	18.800 <sup>c</sup>	9.775	11.248	7.940	13.853	8.794	12.923	7.812
	(0.257)	(0.826)	(0.516)	(0.811)	(0.410)	(0.745)	(0.512)	(0.862)	(0.093)	(0.636)	(0.508)	(0.790)	(0.310)	(0.720)	(0.375)	(0.800)
Q <sup>2</sup> (12)	8.980	13.615	5.438	14.090	6.297	14.103	6.502	14.401	6.331	20.886 <sup>b</sup>	27.233 <sup>a</sup>	13.701	5.748	15.255	6.937	14.507
	(0.705)	(0.326)	(0.942)	(0.295)	(0.900)	(0.294)	(0.889)	(0.276)	(0.899)	(0.052)	(0.007)	(0.320)	(0.928)	(0.228)	(0.862)	(0.270)

Table 5. Cont.

Notes: The number of lags for VAR was decided using SIC (Schwartz information criterion) and AIC (Akaike information criterion) criteria. JB, Q(12) and  $Q^2(12)$  indicate the empirical statistics of the Jarque–Bera test for normality, while Ljung–Box Q statistics of order 12 for autocorrelation were applied to the standardized residuals and squared standardized residuals, respectively. USA, United States of America; IND, India; INDO, Indonesia; KOR, South Korea; MYS, Malaysia; PAK, Pakistan; PHL, the Philippines; TAIW, Taiwan; THA, Thailand. Values in parentheses are the *p*-values. <sup>a</sup>, <sup>b</sup>, <sup>c</sup> indicate the statistical significance at 1%, 5%, and 10%, respectively.

	IND	USA	INDO	USA	KOR	USA	MYS	USA	PAK	USA	PHL	USA	TAIW	USA	THA	USA
							Panel A	A: Mean Equa	tion							
Constant	$3.17 \times 10^{-4}$ (0.227)	$5.16 \times 10^{-4 \text{ b}}$ (0.020)	$-8.21 \times 10^{-5 a}$ (0.743)	$4.52 \times 10^{-4 \text{ b}}$ (0.047)	$2.08 \times 10^{-4}$ (0.395)	$5.26 \times 10^{-4 \text{ b}}$ (0.013)	$-5.44 \times 10^{-5}$ (0.721)	$4.01 \times 10^{-4 \text{ b}}$ (0.055)	$3.33 \times 10^{-4}$ (0.225)	$4.55 \times 10^{-4 \text{ b}}$ (0.030)	$-3.09 \times 10^{-4}$ (0.341)	$4.77 \times 10^{-4 \text{ b}}$ (0.029)	$-3.28 \times 10^{-6}$ (0.988)	$5.05 \times 10^{-4 \text{ b}}$ (0.011)	$1.19 \times 10^{-4}$ (0.557)	$5.36 \times 10^{-4 a}$ (0.005)
$r_{t-1}^o$	0.077 <sup>b</sup> (0.032)	-0.044 (0.166)	0.107 <sup>a</sup> (0.001)	-0.028 (0.359)	-0.020 (0.560)	-0.018 (0.588)	0.089 <sup>b</sup> (0.013)	0.022 (0.685)	0.284 <sup>a</sup> (0.000)	-0.020 (0.367)	0.032 (0.342)	-0.032 (0.188)	0.031 (0.359)	0.076 (0.120)	0.147 <sup>a</sup> (0.000)	-0.024 (0.519)
$r_{t-1}^u$	0.233 <sup>a</sup> (0.000)	-0.082 <sup>b</sup> (0.024)	0.241 <sup>a</sup> (0.000)	-0.069 <sup>c</sup> (0.065)	0.351 <sup>a</sup> (0.000)	-0.079 <sup>b</sup> (0.016)	0.234 <sup>a</sup> (0.000)	-0.070 <sup>c</sup> (0.080)	0.088 <sup>b</sup> (0.016)	-0.077 <sup>b</sup> (0.044)	0.416 <sup>a</sup> (0.000)	-0.077 <sup>b</sup> (0.046)	0.405 <sup>a</sup> (0.000)	-0.081 <sup>b</sup> (0.030)	0.148 <sup>a</sup> (0.000)	-0.054 (0.173)
							Panel B:	Variance Equ	ation							
Constant	$3.75 \times 10^{-6 b}$ (0.020)	$4.42 \times 10^{-6 a}$ (0.000)	$2.05 \times 10^{-6 a}$ (0.005)	4.16 × 10 <sup>-6 a</sup> (0.000)	$1.11 \times 10^{-5 a}$ (0.001)	$-1.41 \times 10^{-6}$ (0.670)	$3.61 \times 10^{-7 \text{ c}}$ (0.092)	$3.28 \times 10^{-6 a}$ (0.004)	$4.09 \times 10^{-6 a}$ (0.000)	4.13 × 10 <sup>-6 a</sup> (0.000)	$4.62 \times 10^{-6 \text{ b}}$ (0.025)	$1.00 \times 10^{-6}$ (0.714)	$2.60 \times 10^{-6 a}$ (0.008)	3.85 × 10 <sup>-6 a</sup> (0.000)	$1.12 \times 10^{-6 a}$ (0.001)	$7.03 \times 10^{-7}$ (0.518)
$\left(e_{t-1}^{o}\right)^{2}$	-0.034 <sup>b</sup> (0.037)	0.079 <sup>a</sup> (0.000)	$2.40 \times 10^{-3}$ (0.911)	0.086 <sup>a</sup> (0.003)	0.149 <sup>a</sup> (0.001)	0.073 (0.270)	0.088 <sup>a</sup> (0.000)	0.018 (0.110)	-0.018 (0.402)	0.011 (0.425)	0.032 (0.277)	0.063 (0.130)	-0.020 (0.271)	0.071 (0.160)	-0.023 (0.146)	0.020 (0.311)
$\left(e_{t-1}^{u}\right)^{2}$	1.73 × 10 <sup>-3</sup> (0.915)	0.051 (0.102)	$-6.47 \times 10^{-4}$ (0.950)	0.065 <sup>c</sup> (0.064)	$9.22 \times 10^{-3}$ (0.738)	0.020 (0.534)	-0.077 <sup>a</sup> (0.000)	0.046 (0.216)	$4.99 \times 10^{-3}$ (0.296)	0.015 (0.640)	-0.014 (0.145)	0.037 (0.343)	0.030 (0.162)	0.068 <sup>c</sup> (0.059)	0.061 <sup>b</sup> (0.046)	0.014 (0.650)
$h_{t-1}^o$	0.896 <sup>a</sup> (0.000)	-0.066 <sup>a</sup> (0.000)	0.891 <sup>a</sup> (0.000)	-0.058 (0.202)	0.511 <sup>a</sup> (0.000)	0.049 (0.243)	0.865 <sup>a</sup> (0.000)	$-7.32 \times 10^{-3}$ (0.556)	0.856 <sup>a</sup> (0.000)	-0.020 (0.155)	0.870 <sup>a</sup> (0.000)	-0.051 <sup>a</sup> (0.004)	0.890 <sup>a</sup> (0.000)	-0.054 <sup>a</sup> (0.001)	0.913 <sup>a</sup> (0.000)	-0.017 (0.219)

Table 6. Estimates of bivariate VAR-AGARCH for American and Asian stock markets during the Chinese Stock Market Crash.

	IND	USA	INDO	USA	KOR	USA	MYS	USA	PAK	USA	PHL	USA	TAIW	USA	THA	USA
							Panel B:	Variance Eq	uation							
$h_{t-1}^u$	-0.012 (0.644)	0.754 <sup>a</sup> (0.000)	0.021 (0.306)	0.696 <sup>a</sup> (0.000)	0.204 (0.149)	0.672 <sup>a</sup> (0.000)	0.303 <sup>a</sup> (0.002)	0.642 <sup>a</sup> (0.000)	-0.012 <sup>a</sup> (0.002)	0.797 <sup>a</sup> (0.000)	0.070 (0.166)	0.711 (0.290)	-0.018 (0.515)	0.726 <sup>a</sup> (0.000)	0.157 <sup>c</sup> (0.072)	0.645 <sup>a</sup> (0.000)
Asymmetry	0.126 <sup>a</sup> (0.002)	0.270 <sup>a</sup> (0.000)	0.123 <sup>a</sup> (0.001)	0.314 <sup>a</sup> (0.000)	-0.079 (0.183)	0.346 <sup>a</sup> (0.000)	0.040 (0.231)	0.317 <sup>a</sup> (0.000)	0.253 <sup>a</sup> (0.000)	0.258 <sup>a</sup> (0.000)	0.081 <sup>b</sup> (0.035)	0.338 <sup>a</sup> (0.000)	0.131 <sup>a</sup> (0.001)	0.296 <sup>a</sup> (0.000)	0.164 <sup>a</sup> (0.000)	0.343 <sup>a</sup> (0.000)
						Pan	el C: Constar	nt Condition	al Correlati	on						
<i>p</i> <sup>0,<i>u</i></sup>	0.324 <sup>a</sup> (0.000)		0.123 <sup>a</sup> (0.000)		0.281 <sup>a</sup> (0.000)		0.146 <sup>a</sup> (0.000)		0.059 <sup>a</sup> (0.078)		0.103 <sup>a</sup> (0.001)		0.214 <sup>a</sup> (0.000)		0.224 <sup>a</sup> (0.000)	
							Panel D	: Diagnostic	Tests							
LogL	5580.558		5500.464		5640.723		5895.388		5471.88		5390.25		5605.08		5699.52	
AIC	-13.712		-13.614		-14.028		-14.525		-13.330		-13.434		-13.877		-13.881	
SIC	-13.418		-13.321		-13.734		-14.232		-13.037		-13.140		-13.583		-13.587	
JB	72.640 <sup>a</sup> (0.000)	204.890 <sup>a</sup> (0.000)	58.520 <sup>a</sup> (0.000)	412.790 <sup>a</sup> (0.000)	134.540 <sup>a</sup> (0.000)	168.440 <sup>a</sup> (0.000)	22.680 <sup>a</sup> (0.000)	442.380 <sup>a</sup> (0.000)	30.880 <sup>a</sup> (0.000)	428.120 <sup>a</sup> (0.000)	27.070 <sup>a</sup> (0.000)	422.700 <sup>a</sup> (0.000)	84.480 <sup>a</sup> (0.000)	230.350 <sup>a</sup> (0.000)	68.060 <sup>a</sup> (0.000)	154.390 <sup>a</sup> (0.000)
Q (12)	16.833 (0.156)	8.760 (0.723)	12.383 (0.415)	9.384 (0.670)	4.583 (0.970)	10.350 (0.585)	7.511 (0.822)	9.705 (0.642)	11.341 (0.500)	10.270 (0.592)	9.383 (0.670)	12.352 (0.418)	20.039 <sup>c</sup> (0.066)	13.680 (0.322)	6.178 (0.907)	7.190 (0.845)
Q <sup>2</sup> (12)	7.383 (0.831)	3.713 (0.988)	18.894 <sup>c</sup> (0.091)	8.117 (0.776)	8.144 (0.774)	5.426 (0.942)	8.099 (0.777)	5.282 (0.948)	7.532 (0.821)	5.161 (0.952)	12.838 (0.381)	7.290 (0.838)	7.271 (0.839)	8.802 (0.720)	24.298 <sup>b</sup> (0.019)	5.675 (0.932)

Table 6. Cont.

Notes: The number of lags for VAR was decided using SIC (Schwartz information criterion) and AIC (Akaike information criterion) criteria. JB, Q(12) and  $Q^2(12)$  indicate the empirical statistics of the Jarque–Bera test for normality, while Ljung–Box Q statistics of order 12 for autocorrelation were applied to the standardized residuals and squared standardized residuals, respectively. USA, United States of America; IND, India; INDO, Indonesia; KOR, South Korea; MYS, Malaysia; PAK, Pakistan; PHL, the Philippines; TAIW, Taiwan; THA, Thailand. Values in parentheses are the *p*-values. <sup>a</sup>, <sup>b</sup>, <sup>c</sup> indicate the statistical significance at 1%, 5%, and 10%, respectively.

The return spillover from the USA to all Asian Stock markets was significant during the Chinese crisis. This implies that US stock market prices played an important role in predicting the prices of all Asian stock markets during the Chinese stock market crash. Moreover, the return spillover from Asia to the US stock market was insignificant. The coefficient of past shock was insignificant in the majority of Asian markets except for in India, Korea, and Malaysia. The sensitivity of past own shocks of the USA was insignificant in most of the cases. In addition, the coefficient of past own volatility significantly affected the conditional volatility of all Asian markets.

The conditional volatility of the majority of Asian stock markets (except Malaysia and Thailand) was not significantly affected by the shocks in the US stock market. In addition, past shocks in most of the Asian stock markets (Except in India and Indonesia) did not influence the conditional volatility of the US stock market. The volatility transmission from the USA to most of the Asian stock markets (except Malaysia, Pakistan, and Thailand) was found to be insignificant during the Chinese Crisis. On the other hand, volatility spillover from most of the Asian stock markets to the USA stock market was evidently insignificant.

The asymmetric coefficients of all Asian stock markets (except Korea and Malaysia) were significant and positive, showing that negative news from the US stock market has a greater ability to increase the volatility of all Asian Stock markets as compared to positive news. However, the asymmetric coefficient of the US stock market is significant and positive. Constant conditional correlation was positively significant for all pairs of stock markets. However, CCC was weak in the majority of pairs.

4.2.6. Stock Market Linkages between China and Asia from the Chinese Stock Market Crash

Table 7 reports the return and volatility spillover between Chinese and Asian stock markets during the Chinese stock market crash. There is significant evidence that lagged returns influence the current stock returns of Asian Stock markets (Except in Korea, the Philippines, and Taiwan). This shows the short-term predictability in stock price changes in the Asian stock markets. Moreover, Chinese stock market returns were not affected by their lags during the Chinese stock market crash.

The return spillover was found to be insignificant from China to all Asian markets. However, the return spillover was found to be insignificant from the majority of Asian markets to the Chinese market, except for India and Taiwan, during the Chinese stock market crash. The coefficient of past own shock did not significantly influence the conditional variance of the most of the Asian stock markets except in India, Malaysia, and Thailand. Moreover, the sensitivity to past own shock of the Chinese stock market was insignificant during the Chinese crash. However, the sensitivity of past own volatility was found to be significant for all Asian stock markets.

The conditional volatility of India, Indonesia, Taiwan, and Thailand was significantly affected by the shocks in the Chinese stock market. However, the shocks in the majority of Asian stock markets (except India and the Philippines) did not influence the Chinese stock market. The past volatility of China significantly impacted the conditional volatility of the stock markets of India, Indonesia, Taiwan, and Thailand. However, volatility spillover was not found from most of the Asian stock markets (except India, Taiwan, and Thailand) to the Chinese stock market during the Chinese stock market crash.

The asymmetric coefficients of all Asian stock markets (except Malaysia and the Philippines) were significant and positive, showing that negative news of the US stock market has a greater ability to increase the volatility of Asian stock markets as compared to positive news. Asymmetric coefficients of China were significant and positive in all pairs, demonstrating that negative news for any Asian markets except for India had a greater ability to increase the volatility of Chinese stock markets as compared to positive news during the Chinese crash. Constant conditional correlation was positively significant for all pairs of stock markets, but CCC was weak in the majority of pairs.

	IND	CHN	INDO	CHN	KOR	CHN	MYS	CHN	РАК	CHN	PHL	CHN	TAIW	CHN	THA	CHN
							Panel A:	Mean Equa	ation							
Constant	$3.32 \times 10^{-4}$ (0.168)	$5.74 \times 10^{-5}$ (0.835)	$1.10 \times 10^{-5}$ (0.964)	$1.39 \times 10^{-4}$ (0.615)	$3.28 \times 10^{-4}$ (0.147)	$8.49 \times 10^{-5}$ (0.715)	$-1.35 \times 10^{-5}$ (0.933)	$9.62 \times 10^{-5}$ (0.715)	$4.13 \times 10^{-4}$ (0.145)	$1.16 \times 10^{-4}$ (0.663)	$5.57 \times 10^{-5}$ (0.875)	$1.07 \times 10^{-4}$ (0.699)	$2.52 \times 10^{-4}$ (0.310)	$9.50 \times 10^{-5}$ (0.714)	$1.73 \times 10^{-4}$ (0.359)	$1.56 \times 10^{-4}$ (0.561)
$r_{t-1}^o$	0.167 <sup>a</sup> (0.000)	0.106 <sup>b</sup> (0.022)	0.144 <sup>a</sup> (0.000)	-0.034 (0.500)	0.084 (0.350)	-0.014 (0.715)	0.125 <sup>a</sup> (0.000)	0.062 (0.322)	0.288 <sup>a</sup> (0.000)	-0.012 (0.660)	0.056 (0.141)	$-1.15 \times 10^{-4}$ (0.997)	0.086 (0.280)	0.106 <sup>b</sup> (0.014)	0.155 <sup>a</sup> (0.000)	0.021 <sup>c</sup> (0.620)
$r_{t-1}^c$	-0.029 (0.185)	0.037 (0.356)	-0.012 (0.550)	0.069 (0.330)	-0.019 (0.361)	0.062 <sup>c</sup> (0.077)	0.021 (0.222)	0.053 (0.181)	0.014 (0.452)	0.062 (0.120)	0.033 (0.274)	0.062 (0.113)	$5.37 \times 10^{-3}$ (0.836)	0.021 (0.580)	$-8.57 \times 10^{-4}$ (0.965)	0.070 <sup>b</sup> (0.052)
							Panel B: V	/ariance Equ	uation							
Constant	3.77 × 10 <sup>-6 a</sup> (0.000)	$1.79 \times 10^{-6 \text{ b}}$ (0.012)	4.23 × 10 <sup>-6 c</sup> (0.010)	$-4.44 \times 10^{-8}$ (0.930)	$3.43 \times 10^{-6}$ (0.139)	$9.89 \times 10^{-7}$ (0.425)	$1.42 \times 10^{-6}$ (0.126)	$-2.42 \times 10^{-7}$ (0.768)	$4.07 \times 10^{-6 a}$ (0.000)	$8.60 \times 10^{-7 a}$ (0.005)	$6.97 \times 10^{-6 \text{ b}}$ (0.032)	$1.31 \times 10^{-6}$ (0.259)	$9.22 \times 10^{-6 a}$ (0.000)	3.22 × 10 <sup>-6 b</sup> (0.017)	$2.64 \times 10^{-6 a}$ (0.001)	$-8.98 \times 10^{-7 \text{ b}}$ (0.031)
$\left(e_{t-1}^{o}\right)^{2}$	-0.063 <sup>a</sup> (0.000)	$^{-5.50 \times}_{10^{-3 a}}$ (0.002)	0.052 (0.119)	$5.85 \times 10^{-4}$ (0.901)	0.024 (0.413)	$5.44 \times 10^{-3}$ (0.107)	0.102 <sup>a</sup> (0.003)	$-8.63 \times 10^{-4}$ (0.677)	-0.013 (0.560)	$-2.66 \times 10^{-5}$ (0.986)	0.058 (0.144)	0.018 <sup>b</sup> (0.022)	-0.028 (0.209)	$5.81 \times 10^{-3}$ (0.347)	-0.039 <sup>b</sup> (0.021)	$-2.55 \times 10^{-3}$ (0.364)
$\left(e_{t-1}^{c}\right)^{2}$	0.070 <sup>a</sup> (0.000)	0.035 <sup>b</sup> (0.031)	-0.018 <sup>a</sup> (0.007)	0.012 (0.567)	$8.96 \times 10^{-3}$ (0.618)	0.023 (0.290)	-0.013 (0.780)	$9.71 \times 10^{-3}$ (0.611)	$5.15 \times 10^{-3}$ (0.180)	$8.77 \times 10^{-3}$ (0.656)	0.021 (0.137)	0.020 (0.342)	0.062 <sup>a</sup> (0.008)	0.027 (0.176)	-0.033 <sup>a</sup> (0.000)	0.017 (0.355)
$h^o_{t-1}$	0.880 <sup>a</sup> (0.000)	0.016 <sup>a</sup> (0.000)	0.789 <sup>a</sup> (0.000)	$8.85 \times 10^{-3}$ (0.176)	0.879 <sup>a</sup> (0.000)	$-2.40 \times 10^{-3}$ (0.665)	0.768 <sup>a</sup> (0.000)	$9.26 \times 10^{-3}$ (0.283)	0.841 <sup>a</sup> (0.000)	$-1.18 \times 10^{-3}$ (0.517)	0.824 <sup>a</sup> (0.000)	$^{-9.03}_{10^{-3}}$ (0.137)	0.659 <sup>a</sup> (0.000)	0.021 <sup>b</sup> (0.038)	0.818 <sup>a</sup> (0.000)	0.015 <sup>b</sup> (0.028)
$h_{t-1}^c$	-0.096 <sup>a</sup> (0.000)	0.945 <sup>a</sup> (0.000)	0.039 <sup>b</sup> (0.025)	0.940 <sup>a</sup> (0.000)	-0.014 (0.742)	0.944 <sup>a</sup> (0.000)	0.100 (0.343)	0.934 <sup>a</sup> (0.000)	$-6.00 \times 10^{-3}$ (0.109)	0.949 <sup>a</sup> (0.000)	-0.026 (0.341)	0.943 <sup>a</sup> (0.000)	-0.118 <sup>b</sup> (0.017)	0.944 <sup>a</sup> (0.000)	0.112 <sup>a</sup> (0.000)	0.933 <sup>a</sup> (0.000)
Asymmetry	0.171 <sup>a</sup> (0.000)	0.029 (0.145)	0.163 <sup>a</sup> (0.000)	0.059 <sup>b</sup> (0.021)	0.044 <sup>c</sup> (0.077)	0.045 <sup>c</sup> (0.058)	0.087 (0.112)	0.065 <sup>b</sup> (0.015)	0.263 <sup>a</sup> (0.000)	0.058 <sup>c</sup> (0.010)	0.061 (0.210)	0.048 <sup>b</sup> (0.045)	0.276 <sup>a</sup> (0.000)	0.047 <sup>b</sup> (0.040)	0.229 <sup>a</sup> (0.000)	0.049 <sup>b</sup> (0.029)
						Pa	nel C: Constan	t Condition	al Correlatio	on						
<i>p</i> <sup>0,c</sup>	0.204 <sup>a</sup> (0.000)		0.151 <sup>a</sup> (0.000)		0.282 <sup>a</sup> (0.000)		0.166 <sup>a</sup> (0.000)		0.109 <sup>a</sup> (0.003)		0.179 <sup>a</sup> (0.000)		0.319 <sup>a</sup> (0.000)		0.202 <sup>a</sup> (0.000)	
							Panel D:	Diagnostic	Tests							
LogL	5216.702		5157.83		5258.48		5520.00		5143.34		5026.77		5246.59		5354.32	
AIC	-12.108 -11.814		-12.093 -11.799		-12.437 -12.143		-12.929 -12.635		-11.856 -11.562		-11.866 -11.573		-12.268 -11.975		-12.364	

Table 7. Estimates of bivariate VAR–AGARCH for Chinese and Asian stock markets during the Chinese Stock Market Crash.

	IND	CHN	INDO	CHN	KOR	CHN	MYS	CHN	PAK	CHN	PHL	CHN	TAIW	CHN	THA	CHN
JB	245.340 <sup>a</sup>	365.250 <sup>a</sup>	212.590 <sup>a</sup>	369.930 <sup>a</sup>	325.490 <sup>a</sup>	305.710 <sup>a</sup>	214.270 <sup>a</sup>	342.420 <sup>a</sup>	156.230 <sup>a</sup>	422.300 <sup>a</sup>	216.090 <sup>a</sup>	338.660 <sup>a</sup>	301.330 <sup>a</sup>	269.370 <sup>a</sup>	227.030 <sup>a</sup>	329.200 <sup>a</sup>
	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)	(0.000)
Q (12)	31.405 <sup>a</sup>	16.802	10.948	15.522	27.305 <sup>a</sup>	16.590	15.069	16.940	46.439 <sup>a</sup>	20.284 <sup>c</sup>	26.154 <sup>c</sup>	15.115	13.912	11.123	24.373 <sup>b</sup>	15.957
	(0.002)	(0.157)	(0.533)	(0.214)	(0.007)	(0.166)	(0.238)	(0.152)	(0.000)	(0.062)	(0.010)	(0.235)	(0.306)	(0.518)	(0.018)	(0.193)
Q <sup>2</sup> (12)	55.871 <sup>a</sup>	17.976	47.212 <sup>a</sup>	21.484 <sup>b</sup>	48.551 <sup>a</sup>	17.416	70.480 <sup>a</sup>	21.463 <sup>b</sup>	70.632 <sup>a</sup>	30.028 <sup>a</sup>	50.180 <sup>a</sup>	23.783 <sup>b</sup>	44.260 <sup>a</sup>	16.310	88.957 <sup>a</sup>	31.358 <sup>a</sup>
	(0.000)	(0.116)	(0.000)	(0.044)	(0.000)	(0.135)	(0.000)	(0.044)	(0.000)	(0.003)	(0.000)	(0.022)	(0.000)	(0.177)	(0.000)	(0.002)

Table 7. Cont.

Notes: The number of lags for VAR was decided using SIC (Schwartz information criterion) and AIC (Akaike information criterion) criteria. JB, Q(12) and  $Q^2(12)$  indicate the empirical statistics of the Jarque–Bera test for normality, while Ljung–Box Q statistics of order 12 for autocorrelation were applied to the standardized residuals and squared standardized residuals, respectively. USA, United States of America; IND, India; INDO, Indonesia; KOR, South Korea; MYS, Malaysia; PAK, Pakistan; PHL, the Philippines; TAIW, Taiwan; THA, Thailand. Values in parentheses are the *p*-values. <sup>a</sup>, <sup>b</sup>, <sup>c</sup> indicate the statistical significance at 1%, 5%, and 10%, respectively.

#### 4.3. Optimal Weights and Hedge Ratio Portfolio Implications

Table 8 indicates the optimal weights and hedge ratios for the pairs of Asia-US stock portfolios during the full sample period, US financial crisis, and the Chinese stock market crash.<sup>7</sup> The range of optimal weights is 0.37 for IND/USA to 0.68 for MYS/USA during the period of the full sample, indicating that for a \$1 India-USA portfolio, 37 cents should be invested in Indian stocks and the remaining 63 cents in the US stock market. The average optimal portfolio weights vary from 0.38 for IND/USA to 0.69 for MYS/USA during the US financial crisis and range from 0.37 for PHL/USA to 0.69 for MYS/USA during the Chinese stock market crash. Overall, the optimal weights of US stock in Asia-USA portfolios are higher during the Chinese stock market crash compared to the US financial crisis. This implies that investors should have maintained more US stocks in their portfolio of Asia-USA during the Chinese stock market crash compared to the US financial crisis.

Table 9 presents the optimal weights and hedge ratios for the pairs of Asia-China stock portfolio during the full sample period, US financial crisis, and the Chinese stock market crash.<sup>8</sup> The range of optimal weights is from 0.56 for IND/CHN, KOR/CHN, PHL/CHN to 0.81 for MYS/CHN during the full sample period. The average optimal portfolio weights vary from 0.53 for IND/CHN to 0.90 for MYS/CHN during the US financial crisis and range from 0.52 for PHL/CHN to 0.82 for MYS/CHN during the Chinese stock market crash. Overall, for the majority of Asia-China portfolios, the optimal weights of Chinese stocks were almost equal or higher during the Chinese stock market crash and the US financial crisis. This suggests that portfolio managers and investors should have maintained almost the same investment in Chinese stock in their majority of the portfolio of Asia-China during both the Chinese crash and the US financial crisis.

Table 8 presents the optimal hedge ratios for the pairs of Asia-USA stock portfolio during the full sample period, US financial crisis, and the Chinese stock market crash. Regarding the hedge ratio, the range of average hedge ratio is 0.04 for PAK/USA to 0.27 for IND/USA during the period of the full sample, showing that a long position of \$1 in Pakistani stocks can be hedged for a short position of 4 cents in US stocks. During the US financial crisis, the average optimal hedge ratios varied from 0.08 for PAK/USA to 0.36 for IND/USA. The average optimal hedge ratio ranged from 0.09 for PAK/USA to 0.36 for IND/USA during the Chinese stock market crash. For the majority of pairs of Asia-USA, the hedge ratios were lower in the US financial crisis compared with the Chinese stock market crash. This suggests that few US stocks were required to minimize the risk of Asian stock investors during the US financial crisis as compared to during the Chinese crash.

Table 9 provides the optimal hedge ratios for the pairs of a Asia-China stock portfolio during the full sample period, US financial crisis, and the Chinese stock market crash. The range of average hedge ratio is 0.04 for PAK/CHN to 0.21 for KOR/CHN during the period of the full sample. During the US financial crisis, the average optimal hedge ratios varied from 0.03 for PAK/CHN to 0.32 for KOR/CHN. The average optimal hedge ratio ranged from 0.09 for MYS/CHN to 0.26 for TAIW/CHN during the Chinese stock market crash. Overall, for the Asia-China pairs, the hedge ratio was lower during the Chinese stock market crash compared to the hedge ratios in the US financial crisis. This implies that fewer Chinese stocks were needed to minimize the risk for Asian stock investors during the Chinese stock market crash as compared to during the US crisis.

<sup>&</sup>lt;sup>7</sup> We calculated the optimal weights by using both VAR-GARCH and VAR-AGARCH models, but we reported the optimal weights only from the VAR-AGARCH model for the purpose of brevity.

<sup>&</sup>lt;sup>8</sup> We calculated the optimal weights by using both VAR-GARCH and VAR-AGARCH models, but we reported the optimal weights only from the VAR-AGARCH model for the purpose of brevity.

	IND/USA	INDO/USA	KOR/USA	MYS/USA	PAK/USA	PHL/USA	TAIW/USA	THA/USA			
Full Sample Period											
$w_t^{SU}$	0.37	0.41	0.40	0.68	0.41	0.42	0.43	0.41			
$\beta_t^{\$U}$	0.27	0.13	0.24	0.06	0.04	0.06	0.17	0.17			
US Financial Crisis											
$w_{t}^{SU}$	0.38	0.51	0.54	0.80	0.52	0.57	0.52	0.52			
$\beta_t^{SU}$	0.36	0.18	0.21	0.11	0.08	0.09	0.15	0.22			
Chinese Stock Market Crash											
$w_t^{SU}$	0.46	0.44	0.51	0.69	0.44	0.37	0.49	0.53			
$\beta_t^{\text{SO}}$	0.36	0.15	0.29	0.11	0.09	0.14	0.22	0.22			

Table 8. Optimal	Weights and	l Hedge Ratios	for Asia/USA pairs.

Note:  $w_t^{SU}$  and  $\beta_t^{SU}$  refer to the optimal weights and hedge ratios, respectively.

Table 9. Optimal	Weights and	Hedge Ratios	for Asia/China	pairs.
	0	0-		

	IND/CHN	INDO/CHN	KOR/CHN	MYS/CHN	PAK/CHN	PHL/CHN	TAIW/CHN	THA/CHN			
	Full Sample Period										
$w_t^{SC}$	0.56	0.57	0.56	0.81	0.57	0.56	0.59	0.59			
$\beta_t^{\&C}$	0.15	0.15	0.21	0.09	0.04	0.12	0.20	0.13			
	US Financial Crisis										
$w_t^{SC}$	0.53	0.63	0.68	0.90	0.64	0.64	0.66	0.67			
$\beta_t^{SC}$	0.31	0.24	0.32	0.13	0.03	0.22	0.30	0.19			
	Chinese Stock Market Crash										
$w_t^{SC}$	0.65	0.61	0.66	0.82	0.58	0.52	0.66	0.73			
$\beta_t^{\&C}$	0.17	0.13	0.23	0.09	0.11	0.18	0.26	0.14			

Note:  $w_t^{SC}$  and  $\beta_t^{SC}$  refer to the optimal weights and hedge ratios, respectively.

#### 5. Conclusions

In this paper, we extend the previous work by examining the return and volatility transmissions from the US and China to the eight emerging Asian stock markets including India, Indonesia, Korea, Malaysia, Pakistan, the Philippines, Taiwan, and Thailand during the Chinese stock market crash by using the VAR-AGARCH model. Moreover, we also examine the spillovers during the full sample period and the 2008 US financial crisis to provide comparative insights to investors about whether the impact of the Chinese crash on equity market spillovers is different from the crashes in other sample periods. Lastly, we also estimate the optimal weights and hedge ratios during the full period and all sub-periods.

Our comprehensive analysis reveals that both return and volatility spillover vary across different pairs of stock markets and during financial crises. The findings of return spillover indicate a significant spillover from the USA to Asian stock markets during the full sample period, the US financial crisis, and the Chinese stock market crash. This implies that US stock market prices play an important role in predicting the prices of the majority of Asian stock markets during the full period and all the sub-periods. However, the return spillover is not significant from China to emerging Asian stock markets during the US financial crisis and the Chinese stock market crash, implying that Chinese stock prices cannot be used for predicting the prices of the majority of Asian stock markets during any of the crisis periods in our study.

Our volatility spillover analysis reveals that the volatility was transmitted from the US to the majority of Asian markets during the full sample period and the Chinese stock market crash, but such a conclusion cannot be drawn for during the US financial crisis. This implies that portfolio investors of Asian stock markets could have gotten the maximum benefits of diversification by holding US stocks in their portfolio during the US financial crisis. However, the volatility spillover was transmitted from China to a majority of Asian markets during the full sample period and US financial crisis, but such a conclusion cannot be reached for during the Chinese crash, implying that portfolio investors of Asian stock markets could have gotten the maximum benefits of diversification by holding Chinese stocks in their portfolio during the Chinese stock market crash.

Based on optimal weights results, the weights of the US stocks in the Asia-USA portfolios are higher during the Chinese crash compared to the US financial crisis, implying that investors should keep more US stocks in their portfolio of the Asia-USA stocks during the Chinese stock market crash, compared to the US financial crisis. For the majority of Asia-China portfolios, the optimal weights of Chinese stocks were almost equal or higher during the Chinese stock market crash and the US financial crisis. This suggests that portfolio managers and investors should have maintained almost the same investment in the Chinese stocks in their portfolio of the Asia-China majority pairs during both the Chinese crash and the US financial crisis.

Regarding the hedge ratios, for most of the Asia-USA pairs, the hedge ratios were smaller in the US financial crisis than in the Chinese stock market crash. This suggests that few US stocks were required to minimize the risk for Asian stock investors during the US financial crisis as compared to during the Chinese crash. In contrast, for the Asia-China pairs, the hedge ratio was smaller during the Chinese stock market crash compared to that in the US financial crisis. This implies that fewer Chinese stocks were needed to minimize the risk for Asian stock investors during the Chinese stock market crash as compared to that in the US financial crisis. This implies that fewer Chinese stocks were needed to the US crisis. Overall, our findings provide several important implications for risk management and portfolio diversification that could be useful for investors and for policymakers related to the US and Asian stock markets.

**Author Contributions:** Conceptualization, estimations, formal analysis, original draft preparation I.Y.; Data collection, methodology writing, and review of draft S.A.; review, editing, and funding W.-K.W. All authors have read and agreed to the published version of the manuscript.

Funding: This research has been supported by Air University, Asia University, China Medical University Hospital, The Hang Seng University of Hong Kong, Research Grants Council (RGC) of Hong Kong (project

number 12500915), and the Ministry of Science and Technology (MOST, Project Numbers 106-2410-H-468-002 and 107-2410-H-468-002-MY3), Taiwan.

**Acknowledgments:** The first author gratefully acknowledge Arshad Hassan (department of Management and Social Sciences, Capital University of Science and Technology, Islamabad) for their valuable suggestions. The third author would like to thank Robert B. Miller and Howard E. Thompson for their continuous guidance and encouragement.

Conflicts of Interest: The authors declare no conflict of interest.

#### References

Allen, Katie. 2015. Why is China's stock market in crisis? The Guardian, July 8.

- Allen, David E., Ron Amram, and Michael McAleer. 2013. Volatility spillovers from the Chinese stock market to economic neighbours. *Mathematics and Computers in Simulation* 94: 238–57. [CrossRef]
- Arouri, Mohamed El Hedi, Amine Lahiani, and Duc Khuong Nguyen. 2011. Return and volatility transmission between world oil prices and stock markets of the GCC countries. *Economic Modelling* 28: 1815–25. [CrossRef]
- Arouri, Mohamed El Hedi, Jamel Jouini, and Duc Khuong Nguyen. 2012. On the impacts of oil price fluctuations on European equity markets: Volatility spillover and hedging effectiveness. *Energy Economics* 34: 611–17. [CrossRef]
- Arouri, Mohamed El Hedi, Amine Lahiani, and Duc Khuong Nguyen. 2015. World gold prices and stock returns in China: Insights for hedging and diversification strategies. *Economic Modelling* 44: 273–282. [CrossRef]
- Baele, Lieven. 2005. Volatility spillover effects in European equity markets. *Journal of Financial and Quantitative Analysis* 40: 373–401. [CrossRef]
- Beirne, John, Guglielmo Maria Caporale, Marianne Schulze-Ghattas, and Nicola Spagnolo. 2013. Volatility spillovers and contagion from mature to emerging stock markets. *Review of International Economics* 21: 1060–75. [CrossRef]
- Bollerslev, T. 1990. Modelling the coherence in short-run nominal exchange rates: A multivariate generalized ARCH model. *The review of Economics and Statistics* 72: 498–505. [CrossRef]
- Bouri, Elie. 2015. Return and volatility linkages between oil prices and the Lebanese stock market in crisis periods. *Energy* 89: 365–71. [CrossRef]
- Cheung, Yan-Leung, Yin-Wong Cheung, and Chris C. Ng. 2007. East Asian equity markets, financial crises, and the Japanese currency. *Journal of the Japanese and International Economies* 21: 138–52. [CrossRef]
- Chien, Mei-Se, Chien-Chiang Lee, Te-Chung Hu, and Hui-Ting Hu. 2015. Dynamic Asian stock market convergence: Evidence from dynamic cointegration analysis among China and ASEAN-5. *Economic Modelling* 51: 84–98. [CrossRef]
- Diebold, Francis X., and Kamil Yilmaz. 2009. Measuring financial asset return and volatility spillovers, with application to global equity markets. *The Economic Journal* 119: 158–71. [CrossRef]
- Dutta, Anupam, Elie Bouri, and Md Hasib Noor. 2018. Return and volatility linkages between CO<sub>2</sub> emission and clean energy stock prices. *Energy* 164: 803–10. [CrossRef]
- Engle, Robert F., Giampiero M. Gallo, and Margherita Velucchi. 2012. Volatility spillovers in East Asian financial markets: A MEM-based approach. *Review of Economics and Statistics* 94: 222–23. [CrossRef]
- Forbes, K., and R. Rigobon. 2002. No contagion, only interdependence: Measuring stock market comovements. *Journal of Finance* 57: 2223–61. [CrossRef]
- Fung, Eric S., Kin Lam, Tak-Kuen Siu, and Wing-Keung Wong. 2011. A Pseudo-Bayesian Model for Stock Returns in Financial Crises. *Journal of Risk and Financial Management* 4: 42–72. [CrossRef]
- Gkillas, Konstantinos, Athanasios Tsagkanos, and Dimitrios I. Vortelinos. 2019. Integration and risk contagion in financial crises: Evidence from international stock markets. *Journal of Business Research* 104: 350–65. [CrossRef]
- Glick, Reuven, and Michael Hutchison. 2013. China's financial linkages with Asia and the global financial crisis. *Journal of International Money and Finance* 39: 186–206. [CrossRef]
- Guo, Xu, Michael McAleer, Wing-Keung Wong, and Lixing Zhu. 2017. A Bayesian approach to excess volatility, short-term underreaction and long-term overreaction during financial crises. *North American Journal of Economics and Finance* 42: 346–58. [CrossRef]

- Hammoudeh, Shawkat M., Yuan Yuan, and Michael McAleer. 2009. Shock and volatility spillovers among equity sectors of the Gulf Arab stock markets. *The Quarterly Review of Economics and Finance* 49: 829–42. [CrossRef]
- He, Ling T. 2001. Time variation paths of international transmission of stock volatility—US vs. Hong Kong and South Korea. *Global Finance Journal* 12: 79–93. [CrossRef]
- Huang, Bwo-Nung, Chin-Wei Yang, and John Wei-Shan Hu. 2000. Causality and cointegration of stock markets among the United States, Japan and the South China Growth Triangle. *International Review of Financial Analysis* 9: 281–97. [CrossRef]
- Huo, Rui, and Abdullahi D. Ahmed. 2017. Return and volatility spillovers effects: Evaluating the impact of Shanghai-Hong Kong Stock Connect. *Economic Modelling* 61: 260–72. [CrossRef]
- Huyghebaert, Nancy, and Lihong Wang. 2010. The co-movement of stock markets in East Asia: Did the 1997–1998 Asian financial crisis really strengthen stock market integration? *China Economic Review* 21: 98–112. [CrossRef]
- In Francis, Sangbae Kim, Jai Hyung Yoon, and Christopher Viney. 2001. Dynamic interdependence and volatility transmission of Asian stock markets: Evidence from the Asian crisis. *International Review of Financial Analysis* 10: 87–96.
- Jin, Xiaoye. 2015. Volatility transmission and volatility impulse response functions among the Greater China stock markets. *Journal of Asian Economics* 39: 43–58. [CrossRef]
- Johansson, Anders C., and Christer Ljungwall. 2009. Spillover effects among the Greater China stock markets. *World Development* 37: 839–51. [CrossRef]
- Jouini, Jamel. 2013. Return and volatility interaction between oil prices and stock markets in Saudi Arabia. *Journal* of Policy Modeling 35: 1124–44. [CrossRef]
- Kim, Bong-Han, Hyeongwoo Kim, and Bong-Soo Lee. 2015. Spillover effects of the US financial crisis on financial markets in emerging Asian countries. *International Review of Economics & Finance* 39: 192–210.
- Kroner, Kenneth F., and Victor K. Ng. 1998. Modeling asymmetric comovements of asset returns. *The Review of Financial Studies* 11: 817–44. [CrossRef]
- Kroner, Kenneth F., and Jahangir Sultan. 1993. Time-varying distributions and dynamic hedging with foreign currency futures. *Journal of financial and Quantitative Analysis* 28: 535–51. [CrossRef]
- Lean, Hooi Hooi, Michael McAleer, and Wing-Keung Wong. 2015. Preferences of risk-averse and risk-seeking investors for oil spot and futures before, during and after the Global Financial Crisis. *International Review of Economics and Finance* 40: 204–16. [CrossRef]
- Li, Yanan, and David E. Giles. 2015. Modelling volatility spillover effects between developed stock markets and Asian emerging stock markets. *International Journal of Finance & Economics* 20: 155–77.
- Lin, Boqiang, Presley K. Wesseh Jr., and Michael Owusu Appiah. 2014. Oil price fluctuation, volatility spillover and the Ghanaian equity market: Implication for portfolio management and hedging effectiveness. *Energy Economics* 42: 172–82. [CrossRef]
- Ling, Shiqing, and Michael McAleer. 2003. Asymptotic theory for a vector ARMA-GARCH model. *Econometric Theory* 19: 280–310. [CrossRef]
- Liu, Y. Angela, and Ming-Shiun Pan. 1997. Mean and volatility spillover effects in the US and Pacific-Basin stock markets. *Multinational Finance Journal* 1: 47–62. [CrossRef]
- Malik, Farooq, and Shawkat Hammoudeh. 2007. Shock and volatility transmission in the oil, US and Gulf equity markets. *International Review of Economics & Finance* 16: 357–68.
- McAleer, Michael, Suhejla Hoti, and Felix Chan. 2009. Structure and asymptotic theory for multivariate asymmetric conditional volatility. *Econometric Reviews* 28: 422–40. [CrossRef]
- Michaud, Richard O., Gary L. Bergstrom, Ronald Frashure, and Brian Wolahany. 1996. Twenty years of international equity investing. *The Journal of Portfolio Management* 23: 9–22. [CrossRef]
- Miyakoshi, Tatsuyoshi. 2003. Spillovers of stock return volatility to Asian equity markets from Japan and the US. *Journal of International Financial Markets, Institutions and Money* 13: 383–99. [CrossRef]
- Sadorsky, Perry. 2012. Correlations and volatility spillovers between oil prices and the stock prices of clean energy and technology companies. *Energy Economics* 34: 248–55. [CrossRef]
- Syriopoulos, Theodore, Beljid Makram, and Adel Boubaker. 2015. Stock market volatility spillovers and portfolio hedging: BRICS and the financial crisis. *International Review of Financial Analysis* 39: 7–18. [CrossRef]
- The causes and consequences of China's market crash. 2015. *The Economist*. Available online: https://www.economist.com/news/2015/08/24/the-causes-and-consequences-of-chinas-market-crash (accessed on 24 September 2020).

- Vieito, João Paulo, Wing-Keung Wong, and Zhen-Zhen Zhu. 2015. Could The Global Financial Crisis Improve the Performance of The G7 Stocks Markets? *Applied Economics* 48: 1066–80. [CrossRef]
- Wan, Henry, Jr., and Wing-Keung Wong. 2001. Contagion or inductance? Crisis 1997 reconsidered. *Japanese Economic Review* 52: 372–80. [CrossRef]
- Xiong, Xiong, Yuxiang Bian, and Dehua Shen. 2018. The time-varying correlation between policy uncertainty and stock returns: Evidence from China. *Physica A: Statistical Mechanics and its Applications* 499: 413–19. [CrossRef]
- Yang, Jian, James W. Kolari, and Insik Min. 2003. Stock market integration and financial crises: The case of Asia. Applied Financial Economics 13: 477–86. [CrossRef]
- Yilmaz, Kamil. 2010. Return and volatility spillovers among the East Asian equity markets. *Journal of Asian Economics* 21: 304–13. [CrossRef]
- Yousaf, Imran, and Arshad Hassan. 2019. Linkages between crude oil and emerging Asian stock markets: New evidence from the Chinese stock market crash. *Finance Research Letters* 31: 207–217. [CrossRef]
- Yousaf, Imran, Shoaib Ali, and Wing-Keung Wong. 2020. Return and Volatility Transmission between World-Leading and Latin American Stock Markets: Portfolio Implications. *Journal of Risk and Financial Management* 13: 148. [CrossRef]
- Yousaf, Imran, and Shoaib Ali. 2020. Linkages between gold and emerging Asian stock markets: New evidence from the Chinese stock market crash. *Studies of Applied Economics* 39. [CrossRef]
- Zhou, Xiangyi, Weijin Zhang, and Jie Zhang. 2012. Volatility spillovers between the Chinese and world equity markets. *Pacific-Basin Finance Journal* 20: 247–70. [CrossRef]
- Zhu, Yanjian, Zhaoying Wu, Hua Zhang, and Jing Yu. 2017. Media sentiment, institutional investors and probability of stock price crash: Evidence from Chinese stock markets. *Accounting & Finance* 57: 1635–70.
- Zhu, Zhenzhen, Zhidong Bai, João Paulo Vieito, and Wing-Keung Wong. 2019. The Impact of the Global Financial Crisis on the Efficiency of Latin American Stock Markets. *Estudios de Economía* 46: 5–30. [CrossRef]



© 2020 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (http://creativecommons.org/licenses/by/4.0/).